Center-Based Approximation of a Drifting Distribution

Alessio Mazzetto Brown University

Matteo Ceccarello Andrea Pietracaprina Geppino Pucci University of Padova ALESSIO_MAZZETTO@BROWN.EDU

MATTEO.CECCARELLO@UNIPD.IT ANDREA.PIETRACAPRINA@UNIPD.IT GEPPINO.PUCCI@UNIPD.IT

ELIEZER_UPFAL@BROWN.EDU

Eliezer Upfal Brown University

Editors: Gautam Kamath and Po-Ling Loh

Abstract

We present a novel technique for computing a center-based approximation of a drifting distribution. Given $k \ge 1$ and a stream of data, whose distribution is changing over time, the goal is to compute, at each step, the best k centers representation of the current distribution, despite possibly having only a single sample from the most recent distribution. In data mining, this is traditionally attempted through the sliding-window mechanism, where the analysis is performed on the most recent fixed-size segment of the data. The problems with this approach are twofold: (1) setting the correct window size is challenging; and (2) a fixed window size cannot effectively track changes in the distribution happening at variable speed. In this paper, we propose a new methodology that dynamically adjusts the window size based on the recent drift of the data. The challenge is that it is not possible to explicitly estimate the drift, as we may have only a single data point from each distribution. Our main contribution lies in providing a rigorous mathematical analysis, establishing both an upper bound via a dynamic window size algorithm, and a lower bound that shows the tightness of our approach.

1. Introduction

We study the problem of approximating a drifting distribution using k points. Given a sequence of independent random vectors $(X_t)_{t\in\mathbb{N}}$ from \mathbb{R}^d , where X_t is a single sample from the distribution P_t , the goal is to compute, at each step T, a set of $k \ge 1$ centers that best represent the current distribution P_T using only the samples X_1, \ldots, X_T . This problem has been extensively studied for a fixed distribution P. Given an integer $k \ge 1$ and a real $z \ge 1$, the objective is to represent P using only k points, dubbed *centers*, such that the expectation of the z-th power distance between a point drawn from P to its closest center is minimized. Formally, the goal is to solve (or approximate) the following minimization problem:

$$\min_{\substack{C \subset \mathbb{R}^d:\\|C|=k}} \mathcal{W}_{k,z}(C,P) \text{ with } \mathcal{W}_{k,z}(C,P) = \int_{x \in \mathcal{X}} \min_{c \in C} ||x-c||^z dP(x) , \qquad (1)$$

where $\mathcal{W}_{k,z}(C, P)$ can be regarded as the cost of approximating P using the set of centers C. As an important special case, the optimal center for k = 1 and z = 2 is the mean of the distribution P. Throughout this paper, we refer to the optimization problem (1) as the (k, z)-approximation problem for a distribution P.

© 2025 A. Mazzetto, M. Ceccarello, A. Pietracaprina, G. Pucci & E. Upfal.

The problem has been studied (mostly for z = 2) under the name of *vector quantization* (Bartlett et al., 1998; Linder, 2000, 2002; Levrard, 2013, 2015), or *clustering* of a distribution (Biau et al., 2008; Maurer and Pontil, 2010; Maurer, 2016; Foster and Rakhlin, 2019; Li and Liu, 2021; Bucarelli et al., 2023). Indeed, for a fixed distribution P, the (k, z)-approximation problem is related to the popular problem of (k, z)-clustering of pointsets (Gupta and Tangwongsan, 2008; Huang et al., 2018; Borassi et al., 2020; Cohen-Addad et al., 2021b). Specifically, a (k, z)-approximation of a fixed distribution P can be obtained by (k, z)-clustering of a finite pointset $S \subseteq \mathbb{R}^d$ is equivalent to solving the (k, z)-approximation problem for the uniform distribution over S. Note that for z = 1 and z = 2, this amounts to computing a solution to the well known k-median and k-means problems, respectively.

In this work, we study the (k, z)-approximation problem in the context of streaming data with concept drift that evolves over time. Specifically, the goal is to solve the (k, z)-approximation problem for the distribution P_T at time T based on the input points $\{X_1, \ldots, X_T\}$ observed so far, which have been sampled independently from a sequence of unknown distributions P_1, \ldots, P_T . Since we have only a single sample X_T from P_T , a solution to this problem must rely on previous samples. However, unlike the static case, where the distribution remains unchanged, using a (k, z)clustering solution over *all* the past samples $\{X_1, \ldots, X_T\}$ from a drifting distribution can result in a poor approximation of the distribution P_T . In fact, since the distribution of the data can change over time, it may be the case that only the most recent data is relevant, and the old data should not influence the current solution. In the learning community, this phenomenon is referred to as *concept drift*, which has been identified as "a major problem in online learning caused when a model based on old data fails to correctly reflect the current state of the world" (Babcock et al., 2003).

Several approaches have been proposed to address concept drift, all sharing the underlying principle of assigning greater relevance to more recent points, typically by applying a decaying weight to older data. A particularly popular framework in this context is the *sliding window model* (Braverman, 2016), where at any given time, a solution is computed solely based on the most recent w points, where w is an integer representing a fixed *window size*. The parameter w plays a crucial role in determining the quality of the solution. Intuitively, a large w may include outdated points due to concept drift. Conversely, if w is too small, the solution may be based on an insufficient number of points, leading to poor generalization. Ideally, the window size should be large enough to contain sufficient data for robust analysis, yet small enough to exclude points that are no longer relevant. This optimal window size may vary over time, particularly in cases where the magnitude of concept drift fluctuates.

Recognizing the importance of the window size, the data mining community has developed several adaptive algorithms that dynamically adjust the window size (Bifet and Gavalda, 2007; Deypir et al., 2012), and these techniques have also been used for clustering (Zhu et al., 2010). However, despite their success in practice, a formal theoretical understanding of the selection of the window size and its relation to the quality of the obtained solution remains largely unexplored. The aim of our work is to bridge this gap. Our main contribution lies in providing a rigorous mathematical analysis of the (k, z)-approximation problem with drifting data, establishing both an upper bound through a dynamic window size algorithm, and a matching lower bound for the problem. Our algorithm adapts to changing drift, computing a near-optimal window size at each step.

Our Contribution We address the important question of how to approximate the current distribution of the data from a sequence of samples whose distribution is changing over time. Our work makes the following specific contributions:

- 1. We introduce a formal theoretical model to analyze the (k, z)-approximation problem for data streams with concept drift.
- 2. We introduce a novel technique to dynamically adjust the window size w in response to the drift of the underlying distribution, and provide a theoretical analysis of the quality of the solution (Theorem 2). This is a challenging problem since, as we will see, it is not possible to quantify the drift from the input data. Despite this, we demonstrate that our method can adjust w as effectively as any algorithm with a full knowledge of the drift.
- 3. Through a minimax lower bound, we establish that our approach strikes a near-optimal tradeoff on the window size w for the (k, 2)-approximation problem (Theorem 4). This lower bound also demonstrates that a properly implemented window size strategy can provide a near-optimal adaptation to drift, justifying previous empirical clustering approaches based on this strategy (Zhu et al., 2010).

Furthermore, we provide preliminary experimental evidence of the ability of our algorithm to adapt the window size according to the drift of the distribution.

Organization of the paper The rest of the paper is structured as follows. Section 2 defines formally the problem setting and outlines the challenges in adapting to drift. Section 3 discusses our main theoretical contributions. Section 4 provides an account of the relevant previous work. Section 5 presents the algorithm and sketches its analysis, while Section 6 presents a sketch of the lower bound. Section 7 reports the results of the experiments. Section 8 concludes with some final remarks. Details of the proofs are deferred to the appendix.

2. Preliminaries

For a point $p \in \mathbb{R}^d$ and a set $S \subseteq \mathbb{R}^d$, we abuse notation and define the distance from p to Sas $||p - S|| \doteq \inf_{s \in S} ||p - s||$, where $|| \cdot ||$ is the Euclidean norm. Given an integer $k \ge 1$, a real $z \ge 1$, and a set of k centers $C \subset \mathbb{R}^d$, we remind that the definition of the cost function $W_{k,z}(C, P)$ for the (k, z)-approximation problem is given by Equation (1). We also recall that the (k, z)-approximation of a distribution P requires finding a set C of at most k points minimizing $W_{k,z}(C, P)$. Also, the (k, z)-clustering problem of a finite multiset of points $S \subseteq \mathbb{R}^d$ is equivalent to the (k, z)-approximation problem of the uniform distribution over S.

We adopt the following data-generating model that has also been used in previous work on drift (Mohri and Muñoz Medina, 2012; Mazzetto and Upfal, 2023b). The input data stream is a sequence of random variables $(X_t)_{t \in \mathbb{Z}^+}$ where X_t , with $t \ge 1$, is sampled from a distribution P_t over \mathbb{R}^d , which represents the current state of the world at time step t. We assume that the random variables $\{X_t : t \in \mathbb{Z}^+\}$ are mutually independent, and that the support of each distribution is contained in the ball of radius 1/2 centered in the origin (i.e., $\Pr(||X_t|| \le 1/2) = 1$ for all $t \ge 1$)¹.

Let C be the family of all possible sets of k centers, that is,

$$\mathcal{C} = \left\{ \{c_1, \dots, c_k\} : c_i \in \mathbb{R}^d, \|c_i\| \le 1/2, \text{ for } 1 \le i \le k \right\} .$$

^{1.} Note that by a simple rescaling, our results can be extended to any finite radius $r \in \mathbb{R}$.

We let $\mathcal{W}_{k,z}^*(P) \doteq \inf_{C \in \mathcal{C}} \mathcal{W}_{k,z}(C, P)$ be the minimum cost attainable for the (k, z)-approximation of P. Let C^* be any optimal solution for P with $\operatorname{cost} \mathcal{W}_{k,z}^*(P)$. We remark that the value $\mathcal{W}_{k,z}^*(P)$ can also be interpreted as the quantization error of representing the distribution P with the points C^* (Graf and Luschgy, 2007; Gersho and Gray, 2012). For instance, if we want to represent the distribution P using a single point (k = 1) with respect to the squared error (z = 2), then C^* is the mean of P, and $\mathcal{W}_{k,z}^*(P)$ is the trace of the covariance matrix of P.

Let T denote the current time. Our goal is to find a set of centers $C \in C$ which minimizes $W_{k,z}(C, P_T)$, thus providing an optimal solution to the (k, z)-approximation problem with respect to the current "state of the world" distribution P_T . One of the main challenges is that the distribution P_T is unknown, thus we cannot directly compute the costs $\{W_{k,z}(C, P_T) : C \in C\}$, needed for the solution of the minimization problem. Since we have only *one single* sample X_T from P_T , those costs must be estimated from past samples, provided that no or minimal drift occurred, i.e., the distribution of those samples closely resembles P_T . Nevertheless, accurately estimating drift is unfeasible, as one sample per distribution is available.

To further proceed with the discussion, we introduce additional notation. To quantify the drift, we define the following problem-specific distance τ between distributions. Given two distributions P and Q over \mathbb{R}^d , we let

$$\tau(P,Q) = \sup_{C \in \mathcal{C}} |\mathcal{W}_{k,z}(C,P) - \mathcal{W}_{k,z}(C,Q)| \quad .$$
⁽²⁾

It is easy to verify that τ defines a pseudometric². In the domain adaptation literature (Mohri and Muñoz Medina, 2012), the distance $\tau(P,Q)$ from P to Q is referred to as the *discrepancy* induced by the family of functions $\{x \mapsto ||x - C||^2 : C \in C\}$. The discrepancy provides a measure of the drift that is problem-dependent. It is possible to upper bound the discrepancy τ using more common metrics between distributions such as the Wasserstein distance (see Appendix A.2).

Let δ_x be the Dirac function centered at $x \in \mathbb{R}^d$. For a window size $1 \le w \le T$, we define

$$P_T^{[w]} = \frac{1}{w} \sum_{t=T-w+1}^T P_t, \qquad \mathbb{P}_T^{[w]} = \frac{1}{w} \sum_{t=T-w+1}^T \delta_{X_t} \quad . \tag{3}$$

The distribution $P_T^{[w]}$ is the average of the distributions of the most recent w samples. The distribution $\mathbb{P}_T^{[w]}$ is the empirical distribution induced by the latest w samples, namely, the discrete distribution where each point $x \in \mathbb{R}^d$ has probability c_x/w , where c_x is the number of realizations of X_{T-w+1}, \ldots, X_T equal to x.

Our goal is to identify a window size w for which the empirical distribution of the samples $\mathbb{P}_T^{[w]}$ is representative of P_T , that is, $\tau(P_T, \mathbb{P}_T^{[w]})$ is small, thus ensuring that the empirical costs computed from the last w samples closely approximate the expected costs of solutions under the true distribution P_T . If we can find such a window size w, then we can obtain a solution for P_T by using an algorithm for (k, z)-clustering over the most recent w samples. To evaluate a window size w, we upper bound $\tau(P_T, \mathbb{P}_T^{[w]})$ based on an error decomposition into *statistical error* and *drift error*, which was used in previous work on drift (Mazzetto and Upfal, 2023a,b). The bound is stated in the following lemma (proof in Appendix B).

^{2.} For a pseudometric τ , it could be that $\tau(x, y) = 0$ even if $x \neq y$.

Lemma 1 For any $1 \le w \le T$, the following inequality holds

$$\tau(P_T, \mathbb{P}_T^{[w]}) \leq \underbrace{\tau(P_T^{[w]}, \mathbb{P}_T^{[w]})}_{statistical \ error} + \underbrace{\max_{0 \leq t < w} \tau(P_T, P_{T-t})}_{drift \ error} \quad .$$

The statistical error quantifies the error caused by the stochasticity of the data-generating process, while the drift error is due to the non-stationarity of the data-generating process. Intuitively, there is a trade-off between those two errors: larger window sizes w yield smaller statistical errors but may incur larger drift errors since we use samples from distributions that are further away from the current distribution.

There are different challenges in the solution of this trade-off. As previously noted, we cannot estimate the drift error since we only have access to a single sample from each distribution. Thus, it is not possible to analytically solve the trade-off to determine the optimal window size. Additionally, it is computationally hard to calculate the metric τ from the data. The computation of the distance τ between two empirical distributions requires evaluating the supremum over all possible sets of k centers in C. This is a computationally hard problem: for k = 1 and z = 1, this is equivalent to the Fermat-Weber problem with attraction and repulsion, for which only approximate solutions are available (e.g., Brimberg (1995)), and no solutions are known for k > 1.

A further obstacle is that even if we can determine the window size w to use, there is no efficient algorithm for computing an optimal solution to the (k, z)-clustering problem (this problem is known to be NP-hard at least for $z \in \{1, 2\}$, see the work of Dasgupta (2008) and Cohen-Addad et al. (2022)). Instead, we assume that a β -approximation algorithm to the (k, z)-clustering problem is available. Such an algorithm computes a solution whose cost is bounded by β times the cost of the optimal solution to the problem, where $\beta \ge 1$ is also referred to as *approximation factor*. We aim to ensure that, despite the algorithm facing multiple complex decisions to tackle drift, we can still achieve the same multiplicative error of the best-known approximation for the considered (k, z)-clustering problem in the no-drift scenario.

3. Main Result

Our work has two primary contributions. Firstly, we show that there exists an algorithm that provides a solution whose cost guarantee is nearly as good as that of a solution computed using a window size that minimizes the trade-off between statistical error and drift error (Theorem 2). Secondly, we exhibit a lower bound for the special case of k-means (z = 2) that demonstrates that our algorithm provides a guarantee that is near-optimal in a drift setting (Theorem 4).

3.1. Upper Bound.

The next theorem characterizes our algorithm's effectiveness in adapting to drift.

Theorem 2 Let $\delta \in (0, 1)$, and assume that we have access to a β -approximation algorithm for the (k, z)-clustering problem. There exists an algorithm that observes X_1, \ldots, X_T and outputs a solution \hat{C} such that, with probability at least $1 - \delta$, it holds

$$\mathcal{W}_{k,z}(\hat{C}, P_T) \le \beta \mathcal{W}_{k,z}^*(P_T) + \beta \cdot O\left(\min_{1 \le w \le T} \left[\sqrt{\frac{z^2 k \log^4(w) + \log(1/\delta)}{w}} + \max_{0 \le t < w} \tau(P_T, P_{T-t})\right]\right)$$

To appreciate this bound, suppose that we fix a window size $w \in \{1, \ldots, T\}$ arbitrarily, and that we compute a β -approximation solution $\tilde{C}_T^{[w]}$ for the (k, z)-clustering problem over the latest w samples X_{T-w+1}, \ldots, X_T , i.e., $\mathcal{W}_{k,z}(\tilde{C}_T^{[w]}, \mathbb{P}_T^{[w]}) \leq \beta \mathcal{W}_{k,z}^*(\mathbb{P}_T^{[w]})$. The following lemma (proof in Appendix **B**) provides an upper bound on the cost of such a solution with respect to the current distribution P_T .

Lemma 3 For any $1 \le w \le T$, the following inequality holds

$$\mathcal{W}_{k,z}(\tilde{C}_T^{[w]}, P_T) \le \beta \cdot \mathcal{W}_{k,z}^*(P_T) + (\beta + 1) \left[\underbrace{\tau(P_T^{[w]}, \mathbb{P}_T^{[w]})}_{statistical \ error} + \underbrace{\max_{0 \le t < w} \tau(P_T, P_{T-t})}_{drift \ error}\right]$$

The upper bound of Lemma 3 is interpreted as follows. The first term $\beta \cdot W_{k,z}^*(P_T)$ represents the multiplicative error introduced by using a β -approximation algorithm, and it is independent to w (we remind that $W_{k,z}^*(P_T)$ is the cost of the optimal solution for P_T). The other term depicts the additive error introduced by the trade-off between the statistical error and drift error (see Lemma 1).

The tightest upper bound of Lemma 3 is achieved by choosing a window size w that optimally solves the trade-off between statistical error and drift error. Theorem 2 shows that we can successfully approach the optimal solution of this trade-off (min in the upper bound), where the statistical error is upper bounded by $\tilde{O}(\sqrt{z^2k/w})$. This upper bound to the statistical error can be proven to be tight up to logarithmic factors (Bartlett et al., 1998) for the k-means problem (z = 2). The additional $O(\sqrt{\log(1/\delta)/w})$ term is due to the high-probability guarantee. We remark that our algorithm does not require any prior knowledge of the drift, and it overcomes the challenge that the drift error *cannot* be estimated from the input data.

Additionally, the proof of Theorem 2 also requires a novel upper bound on the statistical error of the (k, z)-approximation problem for arbitrary $z \ge 1$, extending existing results that only hold for constant z (Bucarelli et al., 2023).

Significantly, our result also features a multiplicative error term proportional to exactly β , with β being the best approximation algorithm for the (k, z)-clustering problem. Although our algorithm involves multiple steps and invocations of a β -approximation algorithm, the final approximation factor remains unaffected.

We emphasize that our algorithm dynamically chooses the window size in order to minimize the trade-off between statistical error and drift error. If the distribution changes arbitrarily and drifts significantly every few steps, learning becomes infeasible, and even the optimal trade-off would yield a poor solution. Also, note that our algorithm makes no prior assumptions about the structure of the drift and can handle arbitrary changes in the distribution. An important special case is gradual drift, where the change in distribution between consecutive steps is bounded; this case is discussed in more detail in Section 3.2. Another important case is abrupt drift, where the input sequence can be divided into long distinct segments, with the distribution remaining constant within each segment but potentially varying significantly across different segments. In such cases, the optimal solution of the trade-off between statistical error and drift error, as outlined in Theorem 2, would lead to selecting samples only from the latest segment. A visual representation of the window sizes selected by the algorithm across these two different drift regimes is provided in Section 7.

3.2. Lower Bound.

We show the tightness of our upper bound for the classical (k, 2)-approximation problem (z = 2). We consider a classical setting where there is a bounded drift at each step, which is widely used in the literature on learning with drift (Bartlett, 1992; Mohri and Muñoz Medina, 2012). Formally, we assume that there exists $\Delta > 0$ such that $\tau(P_{t+1}, P_t) \leq \Delta$ for any $t \geq 1$, which immediately implies that $\max_{t < w} \tau(P_T, P_{T-t}) \leq w\Delta$. Thus, by choosing $w = (k/\Delta^2)^{1/3}$ in Theorem 2, we have that with high-probability (e.g., ≥ 0.99), our algorithm returns a solution \hat{C} such that

$$\mathcal{W}_{k,z}(\hat{C}, P_T) \le \beta \mathcal{W}_{k,z}^*(P_T) + \beta \cdot \tilde{O}\left((k\Delta)^{1/3}\right) \quad . \tag{4}$$

We remark that our algorithm achieves this guarantee without knowing the value Δ a priori.

In our lower bound, we show that the upper bound (4) is tight up to logarithmic factors, and it cannot be improved without further assumptions. The following result follows by combining lower bound techniques for vector quantization (Bartlett et al., 1998) and learning with drift (Mazzetto and Upfal, 2023b).

Theorem 4 Let $k \ge 3$, z = 2, and assume $\Delta \in (0, 1/k^{1+1/d})$ and $T \ge (k^{1-2/d}/\Delta^2)^{1/3}$. Consider an algorithm that observes X_1, \ldots, X_T , and outputs k centers $\hat{C} \in C$. For any such algorithm, there exists a sequence of distributions P_1, \ldots, P_T such that $\tau(P_t, P_{t+1}) \le \Delta$ for any $1 \le t \le T - 1$ and

$$\mathbb{E} \mathcal{W}_{k,z}(\hat{C}, P_T) \ge \mathcal{W}_{k,z}^*(P_T) + \Omega\left(k^{\frac{1}{3}[1-\frac{5}{d}]}\Delta^{1/3}\right) \quad .$$

A sketch of the proof is provided in Section 6, while the full proof is deferred to Appendix C. For $\beta \to 1$ and $d \to \infty$, we observe that the upper bound (4) matches up to logarithmic factors the lower bound in expectation of Theorem 4. This result highlights the tightness of our approach in a drift setting.

We wish to remark that, aside from polylogarithmic factors, there remains a polynomial gap in the dependence on k between the lower and the upper bound, for constant d. This gap is not unique to our analysis, and it also appears in the best-known lower and upper bounds for the setting with independent and identically distributed samples (Bartlett et al., 1998). Indeed, bridging this gap remains an interesting open question.

4. Related Work

The (k, z)-clustering problem, along with its popular variants k-median and k-means, has been extensively studied in the data stream model (see surveys by Silva et al. (2013) and Aggarwal (2018)). A first line of research focused on obtaining good solutions with respect to all the past points in a streaming setting (Ailon et al., 2009; Charikar et al., 2003; Chen, 2006; Braverman et al., 2011; Barger and Feldman, 2016; Cohen-Addad et al., 2023). Different methodologies have been proposed to tackle the concept drift. A line of work developed efficient algorithms that maintain a solution over a sliding window of fixed size (Babcock et al., 2003; Braverman et al., 2015, 2016; Borassi et al., 2020; Epasto et al., 2022; Woodruff et al., 2024). Another alternative is to use a decaying weighting of the past samples (Bidaurrazaga et al., 2021). These works focus on the computational efficiency of maintaining such solutions given an arbitrarily fixed window size or decay parameter. In contrast, we theoretically characterize the quality of the solution for a given window size and propose an *adaptive* method that adjusts the window size based on a simple statistical interpretation

of the data points. While beyond the scope of this paper, it is worth noting that the literature has explored several clustering variants beyond center-based methods (we refer interested readers to the book by Hennig et al. (2015)).

The idea of using sliding windows of adjustable length to adapt to time-varying drift has been empirically explored for clustering (Zhu et al., 2010), and for other data mining problems (e.g., Bifet and Gavalda, 2007; Deypir et al., 2012; Jiang and Zhang, 2004; Kumar and Satapathy, 2014). Unlike these works, we provide a rigorous analysis of the quality of the solution obtained by our algorithm.

There is a vast literature on the problem of concept drift in learning and data mining (Agrahari and Singh, 2022; Gama et al., 2014; Lu et al., 2018). We review the work that is closer to our setting, where the input is modeled as a sequence of independent samples each originating from a different distribution. This drift setting has been widely studied in the learning theory literature, and a first line of work addressed the problem of learning a family of binary classifiers (Helmbold and Long, 1991; Bartlett, 1992; Helmbold and Long, 1994; Barve and Long, 1996; Long, 1998). This analysis was later extended to handle arbitrary family of functions (Mohri and Muñoz Medina, 2012), and to relax the independence assumption using a mixing assumption (Hanneke and Yang, 2019). These works all assume that an upper bound to the magnitude of the drift is known a priori, which is used to quantify the drift error. In the binary and realizable setting, an algorithm that is adaptive to the drift was proposed by Hanneke et al. (2015). More recent work extends this adaptive result to a general learning setting (Mazzetto and Upfal, 2023a). The latter work does not apply to our setting as it requires the computation of the discrepancy (the distance τ defined in (2)), which is computationally unfeasible for the (k, z)-approximation problem. The drift setting has also been considered for other important problems such as density estimation (Mazzetto and Upfal, 2023b; Mazzetto, 2024), weak supervision (Fu et al., 2020; Mazzetto et al., 2025), or model selection (Han et al., 2024).

Another rich area of research for adapting to adversarially changing environments is online learning, where the goal is to minimize regret over a long sequence of predictions with respect to the best solution in hindsight (e.g., see Shalev-Shwartz et al. (2012); Hazan et al. (2016); Orabona (2019)). Recently, the regret of the k-means problem has been studied in this setting (Cohen-Addad et al., 2021a), where at each time step t, the adversary can choose the point X_t after the algorithm determines a solution C_t .

Finally, it is important to note that the statistical error of the (k, z)-approximation problem has been widely studied in the literature for the setting of independent and identically distributed samples. A first line of work established upper bounds for the statistical error of the (k, 2)-approximation problem, which is also referred to as vector quantization, or simply k-means (Bartlett et al., 1998; Linder, 2000; Antos et al., 2005; Biau et al., 2008; Fefferman et al., 2016; Liu, 2021; Appert and Catoni, 2021). An almost matching minimax lower bound for the statistical error is also known for the (k, 2)-approximation problem (Bartlett et al., 1998). Subsequent research provided an upper bound to the statistical error of the (k, z)-approximation problem for a constant z (Bucarelli et al., 2023). Additionally, the statistical error in other clustering variants such as kernel k-means (Liu, 2021), spectral clustering (Li and Liu, 2021), and projective clustering (Bucarelli et al., 2023) has also been explored.

5. Sketch of the Upper Bound

In this section, we present the algorithm that attains the guarantee of Theorem 2. The formal analysis and a more detailed discussion are deferred to Appendix B. As previously noted, the major difficulty

is that it is not possible to obtain an accurate estimate of the drift error $\max_{0 \le t < w} \tau(P_T, P_{T-t})$ from the data, as we have access to only one sample from each distribution, and it is computationally intractable to evaluate the metric τ . Thus, our algorithm needs to choose a window size that yields an optimal solution of the trade-off between statistical error and drift error without explicitly estimating the drift error. Another challenge is that the algorithm can only rely on approximate clustering solutions whose cost can deviate from the optimal arbitrarily within their approximation guarantee.

Our algorithm employs a *lazy strategy*, iteratively increasing the window size until evidence of drift is detected. The key intuition is as follows. Let T be the current time, and let \tilde{C} be an approximate solution for the (k, z)-clustering problem over the latest w samples. We compare \tilde{C} with another solution \tilde{C}' obtained using w' > w samples. If there is no drift, the costs $\mathcal{W}_{k,z}(\tilde{C}, \mathbb{P}_T^{[w]})$ and $\mathcal{W}_{k,z}(\tilde{C}', \mathbb{P}_T^{[w]})$ should not differ beyond the approximation error and the statistical error due to finite sample size w. If these costs are similar, we prefer the solution C' since it exhibits a smaller statistical error, and we continue to increase the window size. Conversely, if the costs are significantly different, it indicates that the data are not all sampled from the same distribution, providing evidence that a drift occurred. In this case, we can establish a lower bound on the drift error using w' samples, which is sufficient to stop iterating. On a high-level, Theorem 2 is derived by carefully determining the threshold for comparing the solutions \tilde{C} and \tilde{C}' .

Formally, the algorithm operates as follows. Let T be the current time. Our algorithm considers windows of doubling sizes $1, 2, 4, \ldots$, and we let $w_j = 2^j$ be the window size considered at iteration j, starting from j = 0. There is nothing special about using powers of 2, and this choice is made only for ease of presentation, since our algorithm can be easily extended to consider any power of $\gamma > 1$ Let $\delta \in (0, 1)$ be the failure probability of our algorithm. The algorithm uses the following upper bound $G(w_j)$ to the statistical error with w_j samples: with probability at least $1 - \delta$, it holds that

$$\forall j \in \mathbb{Z}^+, \quad G(w_j) \doteq c \left[\sqrt{\frac{z^2 k \log^4(w) + \log(1/\delta)}{w_j}} \right] \ge \tau(P_T^{[w_j]}, \mathbb{P}_T^{[w_j]}) \quad , \tag{5}$$

where $c \ge 0$ is a universal constant. The above upper bound is obtained by extending existing results on the statistical error known for constant z to an arbitrary finite $z \ge 1$ (Biau et al., 2008; Maurer, 2016; Liu, 2021; Bucarelli et al., 2023) (see Appendix B.1 for details). We remark that our approach does not require the use of this specific upper bound, and it is trivial to adapt the results to any upper bound on the statistical error, as long as the bound is polynomially decreasing with w.

The algorithm has access to a β -approximation algorithm \mathcal{A} for the (k, z)-clustering problem. At the beginning of iteration j, we compute a solution $\mathcal{A}(\mathbb{P}_T^{[w_j]}) \in \mathcal{C}$ for the (k, z)-clustering problem over the points X_{T-w_j+1}, \ldots, X_T , such that $\mathcal{W}_{k,z}(\mathcal{A}(\mathbb{P}_T^{[w_j]}), \mathbb{P}_T^{[w_j]}) \leq \beta \cdot \mathcal{W}_{k,z}^*(\mathbb{P}_T^{[w_j]})$. For technical reasons, we define $\tilde{C}_T^{[w_j]}$ as the solution that minimizes the cost computed with respect to the empirical distribution $\mathbb{P}_T^{[w_j]}$ among all approximate solutions computed by \mathcal{A} on windows of size w_i with $0 \leq i \leq j$. That is, $\tilde{C}_T^{[w_j]} = \operatorname{argmin} \left\{ \mathcal{W}_{k,z}\left(C, \mathbb{P}_T^{[w_j]}\right) : C \in \{\mathcal{A}(\mathbb{P}_T^{[w_i]})\}_{0 \leq i \leq j} \right\}$. Note that since $\beta > 1$, it is possible that a set of centers computed with respect to another smaller window size $w_i < w_j$ provides a better solution for $\mathbb{P}_T^{[w_j]}$. This definition of $\tilde{C}_T^{[w_j]}$ is used to reduce the effect of the error due to the approximation of the algorithm when comparing the solutions across different window sizes.

Algorithm 1: Adaptive (k, z)-approximation at time step T

 $\begin{array}{c} \overbrace{\tilde{C}_{T}^{[w_{0}]} \leftarrow \mathcal{A}(\mathbb{P}_{T}^{w_{0}})}^{\mathcal{E}_{T} \leftarrow (\mathcal{A}(\mathbb{P}_{T}^{w_{0}}))} \\ \mathcal{S} \leftarrow \left\{\mathcal{A}(\mathbb{P}_{T}^{w_{0}})\right\} \\ \textbf{for } j = 1, 2, \dots, \lfloor \log_{2}(T) \rfloor \textbf{do} \\ & \left\{ \begin{array}{c} \mathcal{S} \leftarrow \mathcal{S} \cup \left\{\mathcal{A}(\mathbb{P}_{T}^{w_{j}})\right\} \\ \widetilde{C}_{T}^{[w_{j}]} \leftarrow \operatorname{argmin}_{C \in \mathcal{S}} \mathcal{W}_{k, z} \left(C, \mathbb{P}_{T}^{[w_{j}]}\right) \\ \textbf{for } i = 0, \dots, j - 1 \textbf{do} \\ & \left| \begin{array}{c} \textbf{if } \mathcal{W}_{k, z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{i}]}) - \mathcal{W}_{k, z}(\tilde{C}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{i}]}) \geq 6G(w_{j}) + 2G(w_{i}) \textbf{ then} \\ & \left| \begin{array}{c} \textbf{return } \tilde{C}_{T}^{[w_{j-1}]} \\ \textbf{end} \\ \textbf{end} \\ \end{array} \right. \\ \textbf{end} \\ \textbf{return } \tilde{C}_{T}^{[w_{\lfloor \log_{2}(T) \rfloor}]} \end{array} \right] \end{array}$

The pseudocode of the algorithm is reported in Algorithm 1. The proof of Theorem 2, provided in full in Appendix B, follows the key intuition presented at the beginning of this section. First, we will show that if at iteration j the if condition is false for all $i = 0, \ldots, j - 1$, then the guarantee provided by the solution $\tilde{C}_T^{[w_j]}$ is up to constants as good as the one provided by any of the previous solutions. In this case, our algorithm keeps iterating. Second, we will show that if at iteration j there exists i < j such that the if condition is true, then a sizeable distribution drift occurred from P_{T-w_j+1} to P_T . This is the crucial technical result that allows the algorithm to lower bound the drift without explicitly computing it. In particular, we will show that when this occurs, then $\max_{0 \le t < w_j} \tau(P_T, P_{T-t}) \ge G(w_j)$. Since the drift error is monotonically increasing with w, the drift error will dominate over the statistical error for any $w \ge w_j$: this is sufficient to interrupt the iteration loop of our algorithm. Then, the proof of the theorem will be constructed using those two results.

Running time and memory. At every time T, under the assumption that all points of the stream X_1, X_2, \ldots, X_T are available in memory, Algorithm 1 must compute at most the $O(\log T)$ solutions and $O(\log^2 T)$ costs. For concreteness, let z = 2. If the solutions are computed using k-means++ followed by a constant number of iterations of Lloyd's algorithm (Arthur and Vassilvitskii, 2007), the overall running time would be $O(kT \log T)$. It is important to remark, that when T grows very large, the performance of the algorithm can be drastically improved by limiting the window sizes to a maximum length L, which would then replace T in the time-bound. In this case, it is easy to argue that the upper bound of Theorem 2 stays the same, aside from an extra error term $O(\beta G(L))$, where G(L) bounds the statistical error with L samples. Thus, if we want to ensure that this extra error is $O(\epsilon)$, we can limit the algorithm to consider window sizes up to length $L = G^{-1}(\epsilon/\beta)$. In this case, it is also sufficient to only keep in memory the most recent O(L) samples at each time step. Further performance improvements can be obtained by leveraging the techniques presented by Epasto et al. (2022) and Woodruff et al. (2024) for k-means clustering in the sliding window setting, by maintaining only a sketch of every window, incurring, however, a slight worsening of the approximation.

6. Sketch of the Lower Bound

In this section, we provide the main intuition behind the proof of the lower bound of Theorem 4. The full details of the proof are deferred to Appendix C. Let T and Δ satisfy the assumption of the theorem. We remind that the goal is to show that for any algorithm that takes in input a sequence X_1, \ldots, X_T from \mathbb{R}^d and outputs a set of at most k centers \hat{C} , there exists a sequence of distributions P_1, \ldots, P_T of the random variables X_1, \ldots, X_T such that $\tau(P_t, P_{t+1}) \leq \Delta$ for each $t \in \{1, \ldots, T-1\}$, and

$$\mathbb{E}\mathcal{W}_{k,2}(\hat{C},P_T) - \mathcal{W}_{k,2}^*(P_T) = \Omega\left(k^{\frac{1}{3}[1-\frac{5}{d}]}\Delta^{1/3}\right)$$

We provide a sketch of the proof for the simpler case where d = 1 and k = 3. To establish the lower bound, we reformulate the problem as a decision problem based on two possible distributions of the input sequence X_1, \ldots, X_T .

In particular, we consider distributions with support over four points $\{x_1, x_2, y_1, y_2\} \subseteq \mathbb{R}$, where $x_1 = 1/2$, $x_2 = 1/2 - \gamma$, $y_1 = -1/4$ and $y_2 = -1/4 + \gamma$. Let $\xi \in \{-1, 1\}$. We consider two distributions over $\mathbf{X} = (X_1, \dots, X_T)$ parameterized by ξ : the product distribution $\mathbf{P}^{[1]} = \bigotimes_{t=1}^T P_t^{[1]}$ and the product distribution $\mathbf{P}^{[-1]} = \bigotimes_{t=1}^T P_t^{[-1]}$. Fix $w \in \{1, \dots, T\}$, we let

$$P_t^{[\xi]}(z) = \begin{cases} \frac{1}{4} + \frac{\xi \Delta}{2} [t - T + w]_+ & \text{if } z \in \{x_1, x_2\} \\ \frac{1}{4} - \frac{\xi \Delta}{2} [t - T + w]_+ & \text{if } z \in \{y_1, y_2\} \end{cases}$$

where $[x]_+ \doteq \max\{0, x\}$. For both sequences, the distribution of the first T - W random variables is uniform over $\{x_1, x_2, y_1, y_2\}$. For the last w random variables, if $\xi = 1$, their distribution will have increasingly larger probability mass to $\{x_1, x_2\}$, and vice-versa for $\xi = -1$. It is easy to see that for any t, the drift is upper bounded $\tau(P_t, P_{t+1}) \le \Delta$ (note that all distances are ≤ 1).

By construction, the optimal centers for $P_T^{[1]}$ are $C^{[1]} \doteq \{x_1, x_2, (y_1 + y_2)/2\}$, whereas the optimal centers for $P_T^{[-1]}$ are $C^{[-1]} = \{y_1, y_2, (x_1 + x_2)/2\}$. Additionally, if $\gamma \leq 1/16$ and $w\Delta \leq 1/16$, one can easily verify that for any set of k centers C, there is a $\hat{C} \in \{C^{[-1]}, C^{[1]}\}$ such that $\mathcal{W}_{3,2}(\hat{C}, P_T^{[\xi]}) \leq \mathcal{W}_{3,2}(C, P_T^{[\xi]})$ for $\xi \in \{-1, 1\}$ (To check this, it is sufficient to consider solutions C constructed as follows: partition $\{x_1, x_2, y_1, y_2\}$ into 3 sets, and find the optimal center for each set). Thus, to establish the desired result, we need only to provide a lower bound to the following quantity:

$$\Psi \doteq \inf_{\hat{C}} \sup_{\xi \in \{-1,1\}} \mathbb{E}_{\mathbf{X} \sim \mathbf{P}^{[\xi]}} \left(\mathcal{W}_{3,2}(\hat{C}, P_T^{[\xi]}) - \mathcal{W}_{3,2}^*(P_T^{[\xi]}) \right) \\ = \inf_{\hat{C} \in \{C^{[-1]}, C^{[1]}\}} \max_{\xi \in \{-1,1\}} \mathbb{E}_{\mathbf{X} \sim \mathbf{P}^{[\xi]}} \left(\mathcal{W}_{3,2}(\hat{C}, P_T^{[\xi]}) - \mathcal{W}_{3,2}^*(P_T^{[\xi]}) \right)$$
(6)

Equation (6) shows that we can obtain a lower bound by formulating a decision problem of determining ξ given the observations X_1, \ldots, X_T . Let $\hat{\phi}$ be any estimator that observes X and it decides whether $\xi = 1$ or $\xi = -1$. The optimal (3, 2)-approximation solution C^* for $P_T^{[\xi]}$ has cost $\frac{\gamma^2}{2}(\frac{1}{2}-w\Delta)$, but if this solution is used to approximate $P_T^{[-\xi]}$, the cost is equal to $\frac{\gamma^2}{2}(\frac{1}{2}+w\Delta)$. Therefore, it holds:

$$\Psi = \gamma^2 w \Delta \cdot \inf_{\hat{\phi}} \max_{\xi \in \{-1,1\}} \mathbb{E}_{\mathbf{X} \sim \mathbf{P}^{[\xi]}} \mathbf{1}_{\{\hat{\phi}(\mathbf{X}) \neq \xi\}} \quad .$$
(7)

where $\mathbf{1}_A \in \{0, 1\}$ is an indicator function that is equal to 1 if and only if A is true. We can use an information-theoretic lower bound as Le Cam's lower bound (Lemma 1 of Yu, 1997) to obtain

$$\inf_{\hat{\phi}} \max_{\xi \in \{-1,1\}} \mathbb{E}_{\mathbf{X} \sim \mathbf{P}^{[\xi]}} \mathbf{1}_{\{\hat{\phi}(\mathbf{X}) \neq \xi\}} \ge \frac{1}{2} \left(1 - \frac{1}{2} \| \mathbf{P}^{[1]} - \mathbf{P}^{[-1]} \|_1 \right) \ge \frac{e^{-\mathrm{KL}(\mathbf{P}^{[1]}, \mathbf{P}^{[-1]})}}{4}$$
(8)

where KL denotes the Kullback–Leibler divergence, and the last inequality is folklore (e.g., Tsybakov, 2008). By direct computation, the KL divergence between the two distributions is upper bounded as

$$\operatorname{KL}\left(\boldsymbol{P}^{[1]}, \boldsymbol{P}^{[-1]}\right) = \sum_{t=1}^{T} \operatorname{KL}\left(\boldsymbol{P}^{[1]}_{t}, \boldsymbol{P}^{[-1]}_{t}\right) = O\left(w^{3}\Delta^{2}\right) \quad . \tag{9}$$

By combining Equations (7), (8) and (9), we finally obtain that:

$$\Psi \ge \gamma^2 w \Delta e^{-O\left(w^3 \Delta^2\right)} \quad . \tag{10}$$

/ [1] [1]

Note that the value of w represents a trade-off: for a larger w, we incur a larger error proportional to $\gamma^2 w$ if we decide incorrectly, however, it is easier to take the correct decision since the two distributions drifted apart for w steps. By taking $\gamma = 1/16$ and $w = (1/\Delta)^{2/3}$ to maximize the lower bound of Equation (10), we finally obtain that $\Psi = \Omega(\Delta^{1/3})$.

To generalize this proof to an arbitrary $k \ge 3$, we construct a decision problem that contains k/3 rescaled instances of the decision problem for k = 3 formulated above. These instances must be sufficiently separated in space so that they can be treated independently. The dependency on the dimension d arises from a covering argument, which determines the scaling required to ensure that all the instances fit within a ball of radius 1/2.

A similar construction to formulate a decision problem to lower bound the error of the (k, 2)approximation problem with independent and identically distributed samples was used by Bartlett
et al. (1998). The extension of the lower bound to take into account a drift in the distribution is
inspired by previous work on learning with drift (Mazzetto and Upfal, 2023b).

7. Simulations

We study the behavior of Algorithm 1 with a proof-of-concept implementation, limited to the case z = 2 (related to the k-means objective). We consider synthetic datasets generated using a mixture of four multivariate Gaussians in two dimensions, where drift is implemented by changing the mean and covariance matrix $\Sigma = I_2 \sigma^2$. Initially, each coordinate of the means is sampled uniformly from the interval [-10, 10] (data will be rescaled later). We consider two datasets. SEGMENTED is a dataset where the stream is divided in four segments of length 256, 512, 128, and 256. Each segment maintains σ constant ($\sigma = 1, 2, 1, \text{ and } 0.5$), and the means are moved at distance 10 in a random direction at the beginning of each segment. In this setting, the optimal window length at any time step t extends from t back to the beginning of the segment containing t. In the STEPWISE dataset, the stream is divided in three periods of 4096 points each, with drift occurring at each time step. In the first and last periods, $\sigma = 1$ and at each time step the means of the Gaussians are shifted by 0.001 at each time step. After generating the stream, we rescale all the points so that they are contained in the unit-norm ball. For both datasets, we report the length of the window selected by our algorithm at each time step, chosen among window sizes with geometric ratio $\gamma = 1.1$.



Figure 1: Window length at each time step on synthetic data: SEGMENTS (top) and STEPWISE (bottom).

For the SEGMENTED dataset (Figure 1, top), we observe that our algorithm (solid line) closely tracks the optimal window length (dashed line) across all segments. Vertical bars mark drift points. For the STEPWISE dataset (Figure 1, bottom) the vertical lines mark the boundary between the different periods. In this regime, we expect the window size to grow until we reach a good solution to the trade-off between statistical error and drift error. This behaviour is confirmed in all three segments. As expected, in the first and last segments, where considerable drift occurs at each time step, our algorithm uses short window lengths. The middle segment has a smaller drift but higher variance, resulting in a statistical error higher than the drift error. Therefore, our algorithm selects longer window lengths.

8. Conclusions

We address the (k, z)-approximation problem for a stream of data whose distribution can change over time. We present a variable-size window algorithm that adjusts its window length to provably adapt to the concept drift, and we show the tightness of our approach through a minimax lower bound for the important special case of z = 2. We believe that our framework is of more general interest, beyond the (k, z)-approximation problem, and can be applied to other important data mining and machine learning problems. As an example, it is possible to extend our work to other clustering variants such as spectral clustering or kernel k-means (Schölkopf et al., 1998; Ng et al., 2001; Li and Liu, 2021).

More generally, the framework works for any real-valued family of functions \mathcal{F} , where the cost $\mathcal{W}(f, P) = \int f(x)dP(x)$ is defined as the expectation of the function $f \in \mathcal{F}$, and $\tau(P,Q)$ is the discrepancy between P and Q according to the family \mathcal{F} . (For the (k, z)-approximation problem in this paper, we used $\mathcal{F} = \{x \mapsto ||x - C||^z : C \in C\}$.) There are only two requirements. The first is that one can compute the function with minimum cost, i.e., the empirical risk minimizer over a set of samples, but an approximate solution to this minimization problem also suffices. The second requirement is that we can provide an upper bound to the statistical error, i.e., a sample complexity result for the uniform convergence of the expected costs of \mathcal{F} . Our result improves on previous adaptive work on learning with drift that requires an exact computation of the function with the minimum cost, and also the additional assumption that the computation of the discrepancy τ is tractable (Mazzetto and Upfal, 2023a).

Acknowledgments. We thank the anonymous reviewers for their valuable feedback and suggestions, which helped improve the quality of this work. This work was partially supported by MUR of Italy, under Projects PRIN 2022TS4Y3N - EXPAND, and PNRR CN00000013 (National Centre for HPC, Big Data and Quantum Computing), and by a Kanellakis Fellowship.

References

- Charu C Aggarwal. A survey of stream clustering algorithms. In *Data Clustering*, pages 231–258. Chapman and Hall/CRC, 2018.
- Supriya Agrahari and Anil Kumar Singh. Concept drift detection in data stream mining: A literature review. *Journal of King Saud University Computer and Information Sciences*, 34(10):9523–9540, 2022.
- Nir Ailon, Ragesh Jaiswal, and Claire Monteleoni. Streaming k-means approximation. *Proc. NIPS*, 22, 2009.
- András Antos, László Gyorfi, and Andras Gyorgy. Individual convergence rates in empirical vector quantizer design. *IEEE Transactions on Information Theory*, 51(11):4013–4022, 2005.
- Gautier Appert and Olivier Catoni. New bounds for *k*-means and information *k*-means. *arXiv* preprint arXiv:2101.05728, 2021.
- David Arthur and Sergei Vassilvitskii. k-means++: the advantages of careful seeding. In *Proc. SODA*, pages 1027–1035, 2007.
- Brain Babcock, Mayur Datar, Rajeev Motwani, and Liadan O'Callaghan. Maintaining variance and k-medians over data stream windows. In *Proc. PODS*, pages 234–243, 2003.
- Artem Barger and Dan Feldman. k-means for streaming and distributed big sparse data. In *Proc. SDM*, pages 342–350, 2016.
- Peter L Bartlett. Learning with a slowly changing distribution. In Proc. COLT, pages 243-252, 1992.
- Peter L Bartlett, Tamás Linder, and Gábor Lugosi. The minimax distortion redundancy in empirical quantizer design. *IEEE Transactions on Information Theory*, 44(5):1802–1813, 1998.
- Rakesh D Barve and Philip M Long. On the complexity of learning from drifting distributions. In Proc. COLT, pages 122–130, 1996.
- Gérard Biau, Luc Devroye, and Gábor Lugosi. On the performance of clustering in Hilbert spaces. *IEEE Transactions on Information Theory*, 54(2):781–790, 2008.
- Arkaitz Bidaurrazaga, Aritz Pérez, and Marco Capó. K-means for evolving data streams. In *ICDM*, pages 1006–1011. IEEE, 2021.
- Albert Bifet and Ricard Gavalda. Learning from time-changing data with adaptive windowing. In *Proceedings of the 2007 SIAM international conference on data mining*, pages 443–448. SIAM, 2007.

- Michele Borassi, Alesandro Epasto, Silvio Lattanzi, Sergei Vassilvitskii, and M. Zadimoghaddam. Sliding window algorithms for k-clustering problems. In *Proc. NeurIPS*, 2020.
- Vladimir Braverman. Sliding window algorithms. In M.Y. Cao, editor, *Encyclopedia of Algorithms*, pages 2006–2011. Springer, 2016.
- Vladimir Braverman, Adam Meyerson, Rafail Ostrovsky, Alan Roytman, Michael Shindler, and Brian Tagiku. Streaming k-means on well-clusterable data. In *Proc. SODA*, pages 26–40, 2011.
- Vladimir Braverman, Harry Lang, Keith Levin, and Morteza Monemizadeh. Clustering on sliding windows in polylogarithmic space. In Proc. FSTTCS, 2015.
- Vladimir Braverman, Harry Lang, Keith Levin, and Morteza Monemizadeh. Clustering problems on sliding windows. In Proc. SODA, pages 1374–1390, 2016.
- Jack Brimberg. The Fermat—Weber location problem revisited. *Mathematical Programming*, 71: 71–76, 1995.
- Maria Sofia Bucarelli, Matilde Fjeldsø Larsen, Chris Schwiegelshohn, and Mads Toftrup. On generalization bounds for projective clustering. In *Proc. NeurIPS*, 2023.
- Moses Charikar, Liadan O'Callaghan, and Rina Panigrahy. Better streaming algorithms for clustering problems. In *Proc. STOC*, pages 30–39, 2003.
- Ke Chen. On k-median clustering in high dimensions. In Proc. SODA, pages 1177-1185, 2006.
- Vincent Cohen-Addad, Benjamin Guedj, Varun Kanade, and Guy Rom. Online k-means clustering. In Proc. AISTATS, pages 1126–1134, 2021a.
- Vincent Cohen-Addad, David Saulpic, and Chris Schwiegelshohn. A new coreset framework for clustering. In Proc. STOC, pages 169–182, 2021b.
- Vincent Cohen-Addad, Karthik C S, and Euiwoong Lee. Johnson coverage hypothesis: Inapproximability of k-means and k-median in *l*p-metrics. In *Proc. SODA*, pages 1493–1530. SIAM, 2022.
- Vincent Cohen-Addad, David P Woodruff, and Samson Zhou. Streaming euclidean k-median and k-means with o (log n) space. In *Proc. FOCS*, pages 883–908. IEEE, 2023.
- Sanjoy Dasgupta. The hardness of k-means clustering. Technical Report CS-2007-0890, University of California, San Diego, 2008.
- Mahmood Deypir, Mohammad Hadi Sadreddini, and Sattar Hashemi. Towards a variable size sliding window model for frequent itemset mining over data streams. *Computers & Industrial Engineering*, 63(1):161–172, 2012.
- Alessandro Epasto, Mohammad Mahdian, Vahab Mirrokni, and Peilin Zhong. Improved sliding window algorithms for clustering and coverage via bucketing-based sketches. In *Proc. SODA*, pages 3005–3042, 2022.

- Charles Fefferman, Sanjoy Mitter, and Hariharan Narayanan. Testing the manifold hypothesis. *Journal of the American Mathematical Society*, 29(4):983–1049, 2016.
- Dylan J Foster and Alexander Rakhlin. 1∞ vector contraction for rademacher complexity. *arXiv* preprint arXiv:1911.06468, 6, 2019.
- Daniel Fu, Mayee Chen, Frederic Sala, Sarah Hooper, Kayvon Fatahalian, and Christopher Ré. Fast and three-rious: Speeding up weak supervision with triplet methods. In *Proc. ICML*, pages 3280–3291, 2020.
- João Gama, Indré Žliobaitė, Albert Bifet, Mykola Pechenizkiy, and Abdelhamid Bouchachia. A survey on concept drift adaptation. *ACM Computing Surveys*, 46(4):1–37, 2014.
- David JH Garling. Inequalities: a journey into linear analysis. Cambridge University Press, 2007.
- Allen Gersho and Robert M Gray. Vector quantization and signal compression, volume 159. Springer Science & Business Media, 2012.
- Siegfried Graf and Harald Luschgy. *Foundations of Quantization for Probability Distributions*. Springer, 2007.
- Anupam Gupta and Kanat Tangwongsan. Simpler analyses of local search algorithms for facility location. *arXiv preprint arXiv:0809.2554*, 2008.
- Elise Han, Chengpiao Huang, and Kaizheng Wang. Model assessment and selection under temporal distribution shift. In *Proc. ICML*, 2024.
- Steve Hanneke and Liu Yang. Statistical learning under nonstationary mixing processes. In *Proc. AISTATS*, pages 1678–1686, 2019.
- Steve Hanneke, Varun Kanade, and Liu Yang. Learning with a drifting target concept. In *Proc. ALT*, pages 149–164, 2015.
- Elad Hazan et al. Introduction to online convex optimization. *Foundations and Trends*® in Optimization, 2(3-4):157–325, 2016.
- David P Helmbold and Philip M Long. Tracking drifting concepts using random examples. In *Proc. COLT*, pages 13–23, 1991.
- David P Helmbold and Philip M Long. Tracking drifting concepts by minimizing disagreements. *Machine Learning*, 14:27–45, 1994.
- Christian Hennig, Marina Meila, Fionn Murtagh, and Roberto Rocci. *Handbook of Cluster Analysis*. CRC press, 2015.
- Lingxiao Huang, Shaofeng H-C Jiang, Jian Li, and Xuan Wu. Epsilon-coresets for clustering (with outliers) in doubling metrics. In *Proc. FOCS*, pages 814–825. IEEE, 2018.
- Jin Jiang and Youmin Zhang. A novel variable-length sliding window blockwise least-squares algorithm for on-line estimation of time-varying parameters. *International Journal of Adaptive Control and Signal Processing*, 18(6):505–521, 2004.

- Vikas Kumar and Sangita Rani Satapathy. A novel technique for mining closed frequent itemsets using variable sliding window. In *Proc. IACC*, pages 504–510, 2014.
- Michel Ledoux and Michel Talagrand. *Probability in Banach Spaces: isoperimetry and processes*. Springer Science & Business Media, 2013.
- Clément Levrard. Fast rates for empirical vector quantization. *Electronic Journal of Statistics*, 7: 1716 1746, 2013.
- Clément Levrard. Nonasymptotic bounds for vector quantization in Hilbert spaces. *The Annals of Statistics*, 43(2):592–619, 2015.
- Shaojie Li and Yong Liu. Sharper generalization bounds for clustering. In *Proc. ICML*, pages 6392–6402, 2021.
- Tamás Linder. On the training distortion of vector quantizers. *IEEE Transactions on Information Theory*, 46(4):1617–1623, 2000.
- Tamás Linder. Learning-theoretic methods in vector quantization. In *Principles of nonparametric learning*, pages 163–210. Springer, 2002.
- Yong Liu. Refined learning bounds for kernel and approximate *k*-means. *Proc. NeurIPS*, 34: 6142–6154, 2021.
- Philip M Long. The complexity of learning according to two models of a drifting environment. In *Proc. COLT*, pages 116–125, 1998.
- Jie Lu, Anjin Liu, Fan Dong, Feng Gu, Joao Gama, and Guangquan Zhang. Learning under concept drift: A review. *IEEE Transactions on Knowledge and Data Engineering*, 31(12):2346–2363, 2018.
- Andreas Maurer. A vector-contraction inequality for rademacher complexities. In *Proc. ALT*, pages 3–17, 2016.
- Andreas Maurer and Massimiliano Pontil. *k*-dimensional coding schemes in hilbert spaces. *IEEE Transactions on Information Theory*, 56(11):5839–5846, 2010.
- Alessio Mazzetto. An improved algorithm for learning drifting discrete distributions. In *Proc. AISTATS*, 2024.
- Alessio Mazzetto and Eli Upfal. An adaptive algorithm for learning with unknown distribution drift. In *Proc. NeurIPS*, 2023a.
- Alessio Mazzetto and Eli Upfal. Nonparametric density estimation under distribution drift. In *Proc. ICML*, pages 24251–24270, 2023b.
- Alessio Mazzetto, Reza Esfandiarpoor, Akash Singirikonda, Eli Upfal, and Stephen H Bach. An adaptive method for weak supervision with drifting data. In *Proc. AISTATS*, 2025.
- Mehryar Mohri and Andres Muñoz Medina. New analysis and algorithm for learning with drifting distributions. In *Proc. ALT*, pages 124–138, 2012.

- Mehryar Mohri, Afshin Rostamizadeh, and Ameet Talwalkar. *Foundations of machine learning*. MIT press, 2018.
- Beny Neta. On three inequalities. Computers & Mathematics with Applications, 6(3):301–304, 1980.
- Andrew Ng, Michael Jordan, and Yair Weiss. On spectral clustering: Analysis and an algorithm. *Proc. NeurIPS*, 14, 2001.
- Francesco Orabona. A modern introduction to online learning. *arXiv preprint arXiv:1912.13213*, 2019.
- Bernhard Schölkopf, Alexander Smola, and Klaus-Robert Müller. Nonlinear component analysis as a kernel eigenvalue problem. *Neural computation*, 10(5):1299–1319, 1998.
- Shai Shalev-Shwartz and Shai Ben-David. Understanding Machine Learning: From Theory to Algorithms. Cambridge university press, 2014.
- Shai Shalev-Shwartz et al. Online learning and online convex optimization. *Foundations and Trends*® *in Machine Learning*, 4(2):107–194, 2012.
- Jonathan A Silva, Elaine R Faria, Rodrigo C Barros, Eduardo R Hruschka, André CPLF de Carvalho, and João Gama. Data stream clustering: A survey. *ACM Computing Surveys (CSUR)*, 46(1):1–31, 2013.
- A. Tsybakov. Introduction to nonparametric estimation. In Springer Series in Statistics, 2008.
- Roman Vershynin. *High-Dimensional Probability: An Introduction with Applications in Data Science*, volume 47. Cambridge university press, 2018.
- David Woodruff, Peilin Zhong, and Samson Zhou. Near-optimal k-clustering in the sliding window model. *Proc. NeurIPS*, 2024.
- Bin Yu. Assouad, fano, and le cam. In *Festschrift for Lucien Le Cam: research papers in probability* and statistics, pages 423–435. Springer, 1997.
- Hongbo Zhu, Yaqiang Wang, and Zhonghua Yu. Clustering of evolving data stream with multiple adaptive sliding window. In 2010 International Conference on Data Storage and Data Engineering, pages 95–100. IEEE, 2010.
- Yubo Zhuang, Xiaohui Chen, and Yun Yang. Wasserstein *k*-means for clustering probability distributions. *Proc. NeurIPS*, 2022.

Appendix A. Auxiliary Results

In this section, we provide technical results that are useful to prove the main results of this paper.

A.1. Useful Inequalities

Proposition 5 Let a and b be two real numbers, and let $z \ge 1$. We have:

$$(a+b)^z \le 2^{z-1}(a^z+b^z)$$
.

Proof *This result is folklore. It holds:*

$$(a+b)^{z} = 2^{z} \left(\frac{a}{2} + \frac{b}{2}\right)^{z} \le 2^{z} \left(\frac{a^{z}}{2} + \frac{b^{z}}{2}\right) = 2^{z-1}(a^{z} + b^{z})$$

where the inequality is due to the convexity of $q \mapsto q^z$ for $z \ge 1$.

Proposition 6 (Neta (1980)) Let $z \ge 1$, and let a and b be two real numbers. We have:

$$|a^{z} - b^{z}| \le \max\left(\frac{z}{2}, 1\right) \cdot (a^{z-1} + b^{z-1}) \cdot |a - b|$$

A.2. Relation between discrepancy and Wasserstein distance

Let P and Q be two distributions over \mathbb{R}^d . A joint distribution ϕ over $(\mathbb{R}^d)^2$ is a coupling for two distributions P and Q over \mathbb{R}^d if $\int \phi(x, y) dx = Q(y)$ and $\int \phi(x, y) dy = P(x)$. The Wasserstein distance W(P, Q) between P and Q is defined as

$$\boldsymbol{W}(P,Q) = \inf_{\phi} \iint \|x - y\|\phi(x,y)dxdy \ .$$

We say that a coupling ϕ^* minimizing the above expression is an optimal coupling for P and Q. Intuitively, ϕ^* is an optimal way to transport the probability mass from P to Q with respect to the Euclidean norm. The Wasserstein distance is often used to describe the distance between distributions for the (k, 2)-approximation problem (e.g., Zhuang et al., 2022).

Lemma 7 (Relation between discrepancy and Wasserstein distance) Let P and Q be two distributions over \mathbb{R}^d with support within the ball of radius 1/2 centered at the origin. Let τ be the discrepancy as defined in Equation 2 for the (k, z)-approximation problem. The following inequality holds:

$$\tau(P,Q) \le \max(z,2) \cdot \boldsymbol{W}(P,Q)$$

Proof Let ϕ^* be the optimal coupling for P and Q. Consider any $C \in \mathcal{C}$. We have that

$$\begin{aligned} \mathcal{W}_{k,z}(C,P) &- \mathcal{W}_{k,z}(C,Q) \\ &= \int \|x - C\|^{z} P(x) dx - \int \|y - C\|^{z} Q(y) dy = \iint (\|x - C\|^{z} - \|y - C\|^{z}) \phi^{*}(x,y) dx dy \\ &\leq \iint \max(z/2,1) \left[|\|x - C\| - \|y - C\|| \cdot (\|x - C\|^{z-1} + \|y - C\|^{z-1}) \right] \phi^{*}(x,y) dx dy \\ &\leq 2 \max(z/2,1) \iint |\|x - C\| - \|y - C\|| \phi^{*}(x,y) dx dy \end{aligned}$$

where in the second equality we used the fact that ϕ^* is a coupling, the third inequality follows from Proposition 6, and the last inequality is due to the fact that P and Q have their support in the ball of radius 1/2. For a point y, let y^C be the closest center of C to y. We have that $||x - C|| - ||y - C|| \le ||x - y^C|| - ||y - y^C|| \le ||x - y||$. With a similar strategy, it is also possible to show $||y - C|| - ||x - C|| \le ||x - y||$. Thus, it holds:

$$2\max(z/2,1)\iint |||x - C|| - ||y - C||| \phi^*(x,y) dxdy$$

$$\leq 2\max(z/2,1)\iint (||x - y||) \phi^*(x,y) dxdy$$

$$= \max(z,2) \mathbf{W}(P,Q) \ .$$

We can prove the same upper bound for $W_{k,z}(C, P) - W_{k,z}(C, Q)$, and we can conclude by observing that the inequality holds for any $C \in C$.

A.3. Discrepancy Inequalities

In this subsection, we will prove inequalities involving the discrepancy, including Lemma 1 and Lemma 3. We first need the following auxiliary result.

Proposition 8 For any $1 \le w \le T$, the following inequality holds

$$\tau(P_T, P_T^{[w]}) \le \max_{0 \le t < w} \tau(P_T, P_{T-t})$$
.

Proof We use the definition of τ and $P_T^{[w]}$ and write

$$\begin{aligned} \tau(P_T, P_T^{[w]}) &= \sup_{C \in \mathcal{C}} \left| \mathcal{W}_{k,z}(C, P_T) - \mathcal{W}_{k,z}(C, P_T^{[w]}) \right| = \sup_{c \in \mathcal{C}} \left| \mathcal{W}_{k,z}(C, P_T) - \frac{1}{w} \sum_{t=T-w+1}^T W(C, P_t) \right| \\ &= \sup_{C \in \mathcal{C}} \left| \frac{1}{w} \sum_{t=T-w+1}^T \left[\mathcal{W}_{k,z}(C, P_T) - \mathcal{W}_{k,z}(C, P_t) \right] \right| \\ &\leq \frac{1}{w} \sum_{t=T-w+1}^T \sup_{C \in \mathcal{C}} \left| \mathcal{W}_{k,z}(C, P_T) - \mathcal{W}_{k,z}(C, P_t) \right| = \frac{1}{w} \sum_{0 \le t < w} \tau(P_T, P_{T-t}) , \end{aligned}$$

where the inequality is due to the triangle inequality. We can conclude that

$$\tau(P_T, P_T^{[w]}) \le \frac{1}{w} \sum_{0 \le t < w} \tau(P_T, P_{T-t}) \le \frac{1}{w} \sum_{0 \le t < w} \sup_{t' < w} \tau(P_T, P_{T-t'}) = \sup_{t' < w} \tau(P_T, P_{T-t'}) .$$

Proof [Lemma 1] We apply the triangle inequality and obtain that

$$\tau(P_T, \mathbb{P}_T^{[w]}) \le \tau(P_T^{[w]}, \mathbb{P}_T^{[w]}) + \tau(P_T^{[w]}, P_T)$$
.

We conclude the proof using Proposition 8

Proof [Lemma 3] The following chain of inequalities holds:

$$\mathcal{W}_{k,z}(\tilde{C}_{T}^{[w]}, P_{T}) - \beta \mathcal{W}_{k,z}^{*}(P_{T}) \\
= \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w]}, P_{T}) - \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w]}, \mathbb{P}_{T}^{[w]}) + \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w]}, \mathbb{P}_{T}^{[w]}) - \beta \mathcal{W}_{k,z}^{*}(P_{T}) \\
\leq |\mathcal{W}_{k,z}(\tilde{C}_{T}^{[w]}, P_{T}) - \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w]}, \mathbb{P}_{T}^{[w]})| + \beta \mathcal{W}_{k,z}(\hat{C}_{T}^{[w]}, \mathbb{P}_{T}^{[w]}) - \beta \mathcal{W}_{k,z}^{*}(P_{T}).$$
(11)

Let C^* be an optimal solution with respect to P_T . We have that $\mathcal{W}_{k,z}^*(P_T) = \mathcal{W}_{k,z}(C^*, P_T)$ and $\mathcal{W}_{k,z}(\hat{C}_T^{[w]}, \mathbb{P}_T^{[w]}) \leq \mathcal{W}_{k,z}(C^*, \mathbb{P}_T^{[w]})$. Therefore, we can upper bound (11) as

$$\mathcal{W}_{k,z}(\tilde{C}_T^{[w]}, P_T) - \beta \mathcal{W}_{k,z}^*(P_T)$$

$$\leq |\mathcal{W}_{k,z}(\tilde{C}_T^{[w]}, P_T) - \mathcal{W}_{k,z}(\tilde{C}_T^{[w]}, \mathbb{P}_T^{[w]})| + \beta |\mathcal{W}_{k,z}(C^*, \mathbb{P}_T^{[w]}) - \beta \mathcal{W}_{k,z}(C^*, P_T)|$$

$$\leq (\beta + 1) \cdot \tau(P_T, \mathbb{P}_T^{[w]}) ,$$

where in the last inequality we used the definition of the metric τ . The statement follows by an application of Lemma 1.

Appendix B. Upper Bound (Theorem 2)

In this section, we provide the full analysis of the algorithm that attains the guarantee of Theorem 2.

As a preliminary step, we formally instantiate and prove the upper bound to the statistical error. We remark that this error is due to the fact that we use the empirical distribution $\mathbb{P}_T^{[w]}$ to approximate the cost of a solution with respect to the underlying distribution $P_T^{[w]}$. This error has been extensively studied for the special case of (k, 2)-approximation in the case of independent samples from the same distribution (e.g., Biau et al., 2008; Maurer, 2016; Li and Liu, 2021), and recent work generalize those results for the (k, z)-approximation problem for constant z (Bucarelli et al., 2023). We provide an upper bound to the statistical error with an explicit dependency on z.

Lemma 9 Consider the (k, z)-approximation problem, where $z \ge 1$. Let $\delta \in (0, 1)$ and $1 \le w \le T$. There exists a constant c > 0 such that with probability at least $1 - \delta$, it holds that:

$$\tau(P_T^{[w]}, \mathbb{P}_T^{[w]}) \le c \left[\sqrt{\frac{z^2 k}{w} \log^4(w)} + \sqrt{\frac{\ln(1/\delta)}{w}} \right]$$

The proof of this result is provided in Subsection B.1. The guarantee of our algorithm is conditioned on the event that we obtain a good estimation of the costs with respect to $P_T^{[w_j]}$ by using the empirical distribution $\mathbb{P}_T^{[w_j]}$, for all $j \ge 0$ (we remind that $w_j = 2^j$). This is formalized in the following corollary, which follows by carefully taking a union bound over the events of Lemma 9.

Corollary 10 Let $\delta \in (0, 1)$. There exists a constant c > 0 such that with probability at least $1 - \delta$, for all $j \ge 0$, it holds that:

$$\tau(P_T^{[w_j]}, \mathbb{P}_T^{[w_j]}) \le c \left[\sqrt{\frac{z^2 k}{w_j} \log^4(w_j)} + \sqrt{\frac{\ln(1/\delta) + \log(\log w_j + 1)}{w_j}} \right]$$

Proof Let $\delta_j = (6/\pi^2)\delta/(j+1)^2$. Lemma 9 implies that for the window size w_j , we have that with probability at least $1 - \delta_j$, it holds

$$\tau(P_T^{[w_j]}, \mathbb{P}_T^{[w_j]}) \le c \left[\sqrt{\frac{z^2 k}{w_j} \log^4(w_j)} + \sqrt{\frac{\ln(1/\delta_j)}{w_j}} \right] \\ = c \left[\sqrt{\frac{z^2 k}{w_j} \log^4(w_j)} + \sqrt{\frac{\ln(\pi^2/(6\delta)) + 2\log(\log w_j + 1)}{w_j}} \right]$$

Note that $\sum_{j\geq 0} (j+1)^{-2} \leq \pi^2/6$. By taking a union bound, the above event holds for all $j \geq 0$ with probability at least $1 - \sum_{j\geq 0} \delta_j \geq 1 - \delta$.

The function $G : \mathbb{N} \mapsto \mathbb{R}$ is defined as the right-hand side of Corollary 10 (compare to Equation (5)):

$$G(w) \doteq c \left[\sqrt{\frac{z^2 k}{w} \log^4(w)} + \sqrt{\frac{\ln(1/\delta) + \log(\log w + 1)}{w}} \right]$$

We remind that $\tilde{C}_T^{[w_j]}$ is defined as the solution that minimizes the cost computed with respect to the empirical distribution $\mathbb{P}_T^{[w_j]}$ among all the approximate solutions computed by \mathcal{A} on windows of size w_i with $0 \le i \le j$,

$$\tilde{C}_T^{[w_j]} = \operatorname{argmin}\left\{ \mathcal{W}_{k,z}\left(C, \mathbb{P}_T^{[w_j]}\right) : C \in \{\mathcal{A}(\mathbb{P}_T^{[w_i]})\}_{0 \le i \le j} \right\}.$$
(12)

To prove Theorem 2 we must show that the solution returned by Algorithm 1 exhibits the quality stated in the theorem. As sketched in Section 5, the argument revolves around two lemmas. The first lemma shows that if at iteration j the if condition is false for all $i = 0, \ldots, j - 1$, then the guarantee provided by the solution $\tilde{C}_T^{[w_j]}$ is up to constants as good as the one provided by any of the previous solutions (Lemma 11). If that is the case, we can keep iterating.

Lemma 11 If the following condition holds for all *i* such that $0 \le i \le j$:

$$\mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, \mathbb{P}_T^{[w_i]}) - \mathcal{W}_{k,z}(\tilde{C}_T^{[w_i]}, \mathbb{P}_T^{[w_i]}) \le 6G(w_j) + 2G(w_i)$$
,

then we have that

$$\mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, P_T) \le \beta \cdot \mathcal{W}_{k,z}^*(P_T) + \min_{0 \le i \le j} \left[(\beta + 1) \cdot \tau(P_T, \mathbb{P}_T^{[w_i]}) + 6G(w_j) + 2G(w_i) \right] .$$

Proof Fix an integer *i* such that $0 \le i \le j$. We have that

$$\mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, P_T) - \beta \cdot \mathcal{W}_{k,z}^*(P_T)$$

= $\mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, P_T) - \beta \cdot \mathcal{W}_{k,z}^*(P_T) + \mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, \mathbb{P}_T^{[w_i]}) - \mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, \mathbb{P}_T^{[w_i]})$ (13)

Now observe that

$$\begin{aligned} \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{i}]}) &\leq \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{i}]}) + 6G(w_{j}) + 2G(w_{i}) \\ &\leq \beta \cdot \mathcal{W}_{k,z}^{*}(\mathbb{P}_{T}^{[w_{i}]}) + 6G(w_{j}) + 2G(w_{i}) \\ &\leq \beta \cdot \mathcal{W}_{k,z}(C^{*}, \mathbb{P}_{T}^{[w_{i}]}) + 6G(w_{j}) + 2G(w_{i}) , \end{aligned}$$

where C^* is a solution such that $\mathcal{W}_{k,z}^*(P_T) = \mathcal{W}_{k,z}(C^*, P_T)$. By plugging the above inequality into (13), we obtain that

$$\begin{split} & \mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, P_T) - \beta \cdot \mathcal{W}_{k,z}^*(P_T) \\ & \leq \left(\mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, P_T) - \mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, \mathbb{P}_T^{[w_i]}) \right) \\ & + \beta \left(\mathcal{W}_{k,z}(C^*, \mathbb{P}_T^{[w_i]}) - \mathcal{W}_{k,z}(C^*, P_T) \right) + 6G(w_j) + 2G(w_i) \\ & \leq (\beta + 1) \cdot \tau(\mathbb{P}_T^{[w_i]}, P_T) + 6G(w_j) + 2G(w_i) \enspace . \end{split}$$

The second proposition shows that if at iteration j there exists i < j such that the *if* condition is true, then a sizeable distribution drift occurred from P_{T-w_j+1} to P_T . When this happens, we can interrupt the iteration loop of our algorithm (Lemma 12).

Lemma 12 Assume that the event of Corollary 10 holds. If there exists *i* and *j*, with $0 \le i < j$, such that

$$\mathcal{W}_{k,z}(\tilde{C}_T^{[w_j]}, \mathbb{P}_T^{[w_i]}) - \mathcal{W}_{k,z}(\tilde{C}_T^{[w_i]}, \mathbb{P}_T^{[w_i]}) \ge 6G(w_j) + 2G(w_i)$$

then $\max_{0 \le t < w_j} \tau(P_T, P_{T-t}) \ge G(w_j).$

Proof The proof of this proposition relies on the following inequality

$$\tau(P_T^{[w_j]}, P_T^{[w_i]}) \le \tau(P_T, P_T^{[w_j]}) + \tau(P_T, P_T^{[w_i]}) \le 2 \sup_{t < w_j} \tau(P_T, P_{T-t}) \quad , \tag{14}$$

where the last step is due to Proposition 8.

Let i and j be the indexes defined in the statement of the proposition. We divide the analysis into two cases: (a) when inequality

$$W(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{i}]}) - \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{j}]}) \ge 3G(w_{j}) + G(w_{i})$$
(15)

holds, or (b) when inequality (15) does not hold.

Case (a). We are assuming that (15) holds. In this case, we have that

$$\begin{aligned} \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{i}]}) &- \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{j}]}) \leq \tau(\mathbb{P}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{j}]}) \\ &\leq \tau(\mathbb{P}_{T}^{[w_{i}]}, P_{T}^{[w_{i}]}) + \tau(P_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{j}]}) + \tau(P_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{j}]}) \\ &\leq \tau(P_{T}^{[w_{i}]}, P_{T}^{[w_{j}]}) + G(w_{i}) + G(w_{j}) \end{aligned}$$

The second inequality is obtained by using the triangle inequality, and the third inequality is due to Corollary 10. By combining the above inequality with (14), we obtain that $\tau(P_T^{[w_i]}, P_T^{[w_j]}) \ge G(w_j)$.

Case (b). Inequality (15) does not hold. In this case, we have that:

$$\begin{split} & \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{i}]}) - \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{i}]}) \\ & \leq \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{j}]}) - \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{i}]}) + G(w_{i}) + 3G(w_{j}) \\ & \leq \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{j}]}) - \mathcal{W}_{k,z}(\tilde{C}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{i}]}) + G(w_{i}) + 3G(w_{j}) \\ & \leq \tau(\mathbb{P}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{j}]}) + G(w_{i}) + 3G(w_{j}) \\ & \leq \tau(\mathbb{P}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{i}]}) + \tau(\mathbb{P}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{j}]}) + \tau(\mathbb{P}_{T}^{[w_{j}]}, \mathbb{P}_{T}^{[w_{j}]}) \\ & \leq \tau(\mathbb{P}_{T}^{[w_{i}]}, \mathbb{P}_{T}^{[w_{j}]}) + 2G(w_{i}) + 4G(w_{j}) . \end{split}$$

The first inequality is due to (15), the second inequality is obtained by using the definition of $\tilde{C}_T^{[w_j]}$ (12), and the last inequality is due to Corollary 10. By using the assumption of the proposition, we have that

$$\tau(P_T^{[w_i]}, P_T^{[w_j]}) \ge 6G(w_j) + 2G(w_i) - 4G(w_j) - 2G(w_i) = 2G(w_j) .$$

We can conclude the proof by using inequality (14).

The above lemma provides a lower bound to the drift error when the stopping condition is verified. This is the crucial technical result that allows the algorithm to lower bound the drift without explicitly computing it. The proof of the theorem is constructed with those two lemmas. **Proof** [Theorem 2] Consider the function $\Phi(w)$ defined as

$$\Phi(w) = \beta \left[G(w) + \sup_{t < w} \tau(P_T, P_{T-t}) \right] \quad .$$

Let $w^* = \operatorname{argmin}_{1 \le w \le T} \Phi(w)$ be the optimal window size according to Φ . The algorithm returned $\tilde{C}^{[w_j]}$ for some j such that $0 \le j \le \lfloor \log_2 T \rfloor$. To prove the theorem, it is sufficient to show that

$$\frac{\mathcal{W}_{k,z}(\tilde{C}^{\lfloor w_j \rfloor}) - \beta \cdot \mathcal{W}_{k,z}^*(P_T)}{\Phi(w^*)} = O(1) \quad .$$

We assume that the event of Corollary 10 holds, otherwise we say that our algorithm fails (with probability $\leq \delta$). Let ℓ be the largest integer such that $2^{\ell} \leq w^*$. We distinguish two cases: $\ell \leq j$ and $\ell > j$.

(Case $\ell \leq j$). Since the algorithm returned the value j, we have that Lemma 11 holds for j. As $\ell \leq j$, we obtain

$$\begin{split} & \frac{\mathcal{W}_{k,z}(\tilde{C}^{[w_j]}) - \beta \cdot \mathcal{W}_{k,z}^*(P_T)}{\Phi(w^*)} \\ \leq & \frac{(\beta+1) \cdot \tau(P_t, \mathbb{P}_T^{[w_\ell]}) + 6G(w_j) + 2G(w_\ell)}{\Phi(w^*)} \\ \leq & \frac{(\beta+1) G(w_\ell) + (\beta+1) \sup_{t < w_\ell} \tau(P_T, P_{T-t}) + 6G(w_j) + 2G(w_\ell)}{\beta G(w^*) + \beta \sup_{t < w^*} \tau(P_T, P_{T-t})} \\ \leq & \left(1 + \frac{3}{\beta}\right) \frac{G(w_\ell)}{G(w^*)} + \left(1 + \frac{1}{\beta}\right) \frac{\sup_{t < w^*} \tau(P_T, P_{T-t})}{\sup_{t < w^*} \tau(P_T, P_{T-t})} + \frac{6}{\beta} \frac{G(w_j)}{G(w^*)} \\ = & O(1) \quad , \end{split}$$

where the second inequality is due to Lemma 1.

(Case $\ell > j$). Since the algorithm terminated with $j < \ell$, it means that there exists a value of i such that the iteration of the algorithm terminated with indexes (j + 1, i). Due to Lemma 12, this implies that

$$\sup_{t < w^*} \tau(P_T, P_{T-t}) \ge \sup_{t < w_{j+1}} \tau(P_T, P_{T-t}) \ge G(w_{j+1}) \quad .$$
(16)

We have that

$$\frac{\mathcal{W}_{k,z}(\tilde{C}^{[w_{j}]}) - \beta \cdot \mathcal{W}_{k,z}^{*}(P_{T})}{\Phi(w^{*})} \leq \frac{(\beta+1) G(w_{j}) + (\beta+1) \sup_{t < w_{j}} \tau(P_{T}, P_{T-t})}{\beta G(w^{*}) + \beta \sup_{t < w^{*}} \tau(P_{T}, P_{T-t})} \leq \left(1 + \frac{1}{\beta}\right) \frac{G(w_{j})}{\sup_{t < w^{*}} \tau(P_{T}, P_{T-t})} + \left(1 + \frac{1}{\beta}\right) \frac{\sup_{t < w^{*}} \tau(P_{T}, P_{T-t})}{\sup_{t < w^{*}} \tau(P_{T}, P_{T-t})} \leq \left(1 + \frac{1}{\beta}\right) \frac{G(w_{j})}{G(w_{j+1})} + \left(1 + \frac{1}{\beta}\right) = O(1) .$$

The first inequality follows from Lemma 1, and the third inequality is due to (16).

B.1. Upper bound on the Statistical Error (Lemma 9)

In this subsection, we prove the upper bound on the statistical error of Lemma 9 for the (k, z)-approximation problem. The statistical error of the (k, z)-approximation problem has already been studied by Bucarelli et al. (2023) for arbitrary k and constant z. In particular, we will use their main result for the case $1 \le z < 2$.

Lemma 13 (Bucarelli et al. (2023)) Let $1 \le z < 2$, $w \in \{1, ..., T\}$, and $\delta \in (0, 1)$. There exists a constant *c* such that with probability at least $1 - \delta$, it holds that:

$$\tau(P_T^{[w]}, \mathbb{P}_T^{[w]}) \le c \cdot \left(\sqrt{\frac{k \log^4(w)}{w}} + \sqrt{\frac{\log(1/\delta)}{w}}\right)$$

Proof This result is a straightforward adaptation of Theorem 5.1 in Bucarelli et al. (2023) to our setting.

Lemma 13 shows that Lemma 9 is true for $1 \le z < 2$. In the remaining of this subsection, we will focus on the case $z \ge 2$. Compared to the previous work, our goal is to provide a characterization of the statistical error with an explicit dependency on z. To obtain this result, we extend the argument by Liu (2021) based on the Rademacher complexity for (k, 2)-approximation problem to an arbitrary $z \ge 2$.

Consider the family of functions $\mathcal{F} : \mathbb{R}^d \mapsto \mathbb{R}$ defined as:

$$\mathcal{F} = \{ x \mapsto \| x - C \|^z : C \in \mathcal{C} \}$$

The empirical Rademacher complexity of the family \mathcal{F} (Shalev-Shwartz and Ben-David, 2014) over the last w steps is defined as:

$$\hat{\mathcal{R}}_w(\mathcal{F}) \doteq \mathop{\mathbb{E}}_{\sigma} \sup_{f \in \mathcal{F}} \left(\sum_{t=T-w+1}^T \sigma_t f(X_t) \right) ,$$

where $\sigma_1, \ldots, \sigma_T$ are independent Rademacher random variables, i.e., $\sigma_t = \pm 1$ with probability 1/2. The Rademacher complexity of \mathcal{F} over the previous w steps is $\mathcal{R}_w(\mathcal{F}) \doteq \mathbb{E}_X \hat{\mathcal{R}}_w(\mathcal{F})$. The following result provides an upper bound to the empirical Rademacher complexity, and it will be proven at the end of this subsection.

Lemma 14 Let $z \ge 2$, and let $\mathcal{X} = \{x : ||x||^2 \le 1/2\}$. There exists a constant c > 0 such that for any set of samples $\{X_{T-w+1}, \ldots, X_T\} \in \mathcal{X}^w$, we have that:

$$\hat{\mathcal{R}}_w(\mathcal{F}) \le c \cdot z \sqrt{kw \log^4(w)}$$
.

The above upper bound to the empirical Rademacher complexity can be used to prove an upper bound to the statistical error through classical learning theory techniques (Mohri et al., 2018). **Proof** [Lemma 9] If $1 \le z < 2$, the result immediately follows from Lemma 13. Let $z \ge 2$. We can observe that $||f||_{\infty} \le 1$ for any $f \in \mathcal{F}$ since all the points are contained in the ball of radius 1/2centered in the origin. We can use McDiarmid's inequality, and show that with probability at least $1 - \delta/2$ it holds that:

$$\frac{1}{w}\mathcal{R}_w(\mathcal{F}) \le \frac{1}{w}\hat{\mathcal{R}}_w(\mathcal{F}) + O\left(\sqrt{\frac{\log(1/\delta)}{w}}\right) \quad . \tag{17}$$

We use Lemma 14 to upper bound the empirical Rademacher complexity. Thus, we have that:

$$\frac{1}{w}\mathcal{R}_w(\mathcal{F}) = O\left(z\sqrt{\frac{k}{w}\log^4(w)} + \sqrt{\frac{\ln(1/\delta)}{w}}\right) \quad . \tag{18}$$

Let A be the random variable of the statement of the lemma

$$A = \tau(P_T^{[w]}, \mathbb{P}_T^{[w]}) \ . \tag{19}$$

By a standard symmetrization argument (Vershynin, 2018), we have that

$$\mathbb{E}A \le (2/w)\mathcal{R}_w(\mathcal{F}) \quad . \tag{20}$$

We can use McDiarmid's inequality again and show that with probability at least $1 - \delta/2$, it holds:

$$A \le \mathbb{E} A + O\left(\sqrt{\frac{\ln(1/\delta)}{w}}\right) \quad . \tag{21}$$

By combining (18), (20) and (21), and taking a union bound, we obtain the final statement.

The remaining of this subsection is dedicated to the proof of Lemma 14. The first step is the application of the ℓ_{∞} vector contraction for the Rademacher complexity (Foster and Rakhlin, 2019).

Lemma 15 Let $\mathcal{X} = \{x : ||x||^2 \le 1/2\}$. There exists a constant c > 0 such that for any set of samples $\{X_{T-w+1}, \ldots, X_T\} \in \mathcal{X}^w$, we have that:

$$\hat{\mathcal{R}}_w(\mathcal{F}) \le c\sqrt{k}\tilde{\mathcal{R}}_w(\overline{\mathcal{F}})\log^2\left(\frac{w}{\tilde{\mathcal{R}}_w(\overline{\mathcal{F}})}\right)$$
,

where $\overline{\mathcal{F}} = \{x \mapsto \|x - c\|^z : c \in \mathcal{X}\}$, and $\tilde{\mathcal{R}}_w(\overline{\mathcal{F}}) = \sup_{\{X_{T-w+1}, \dots, X_T\} \in \mathcal{X}^w} \hat{\mathcal{R}}_w(\overline{\mathcal{F}}).$

Proof It is a straightforward adaptation of the proof of Lemma 1 in Liu (2021) to our setting.

Lemma 15 shows that in order to upper bound the empirical Rademacher complexity of \mathcal{F} , it is sufficient to evaluate the worst-case empirical Rademacher complexity $\tilde{\mathcal{R}}_w(\overline{\mathcal{F}})$, where $\overline{\mathcal{F}}$ is a family of functions that is restricted to consider a single center. To this end, we show both an upper bound (Proposition 16) and a lower bound (Proposition 17) to $\tilde{\mathcal{R}}_w(\overline{\mathcal{F}})$.

Proposition 16 (Upper bound to worst-case empirical Rademacher complexity) *The following inequality holds:*

$$\tilde{\mathcal{R}}_w(\overline{\mathcal{F}}) \leq \frac{3z}{4}\sqrt{w}$$
.

Proof Let $\mathcal{X} = \{x \in \mathbb{R}^d : ||x||^2 \le 1/2\}$. Consider an arbitrary input $\{X_{T-w+1}, \ldots, X_T\} \in \mathcal{X}^w$. We have that:

$$\hat{\mathcal{R}}_w(\overline{\mathcal{F}}) = \mathop{\mathbb{E}}_{\sigma} \sup_{f \in \overline{\mathcal{F}}} \left(\sum_{t=T-w+1}^T \sigma_t f(X_t) \right) = \mathop{\mathbb{E}}_{\sigma} \sup_{c \in \mathcal{X}} \left(\sum_{t=T-w+1}^T \sigma_t \|X_t - c\|^z \right) \ .$$

For $t \in \{T - w + 1, ..., T\}$, let $\psi_t : \mathcal{X} \mapsto \mathbb{R}$ be defined as $\psi_t(c) \doteq ||X_t - c||^2$, and let $h : \mathbb{R} \mapsto \mathbb{R}$ be the power function $h(a) = a^{z/2}$. It is possible to observe that $||X_t - c||^2 = h(\psi_t(c))$. Note that $0 \le \psi_t(c) \le 1$ for any t and $c \in \mathcal{X}$, and h is (z/4)-Lipschitz for $a \in [0, 1]$. By using the Lipschitz contraction inequality for Rademacher complexity (Ledoux and Talagrand, 2013), it is possible to show that

$$\hat{\mathcal{R}}_w(\overline{\mathcal{F}}) = \underset{\sigma}{\mathbb{E}} \sup_{c \in \mathcal{X}} \sum_{t=T-w+1}^T \sigma_t h(\psi_t(c)) \le \frac{z}{4} \underset{\sigma}{\mathbb{E}} \sup_{c \in \mathcal{X}} \sum_{t=T-w+1}^T \sigma_t \|X_t - c\|^2 .$$
(22)

We upper bound the expectation of the right-hand side of (22) as follows:

$$\mathbb{E}\sup_{\sigma} \sum_{c \in \mathcal{X}} \sum_{t=T-w+1}^{T} \sigma_t \|X_t - c\|^2 = \mathbb{E}\sup_{\sigma} \sum_{c \in \mathcal{X}} \sum_{t=T-w+1}^{T} \sigma_t \left(\|X_t\|^2 + \|c\|^2 - 2\langle X_t, c \rangle \right) \\
\leq \mathbb{E}\sup_{\sigma} \sum_{c \in \mathcal{X}} \sum_{t=T-w+1}^{T} \sigma_t \|c\|^2 + 2\mathbb{E}\sup_{\sigma} \sum_{c \in \mathcal{X}} \sum_{t=T-w+1}^{T} \sigma_t \langle X_t, c \rangle . \quad (23)$$

We have the following upper bounds:

$$\mathbb{E}\sup_{\sigma} \sup_{c \in \mathcal{X}} \sum_{t=T-w+1}^{T} \sigma_t \|c\|^2 \le \mathbb{E}\sup_{\sigma} \|c\|^2 \left| \sum_{t=T-w+1}^{T} \sigma_t \right| \le \sqrt{w} \quad , \tag{24}$$

where the last step is due to the Khintchine inequality (Garling, 2007). Also, it holds:

$$2 \mathop{\mathbb{E}}_{\sigma} \sup_{c \in \mathcal{X}} \sum_{t=T-w+1}^{T} \sigma_t \langle X_t, c \rangle = 2 \mathop{\mathbb{E}}_{\sigma} \sup_{c \in \mathcal{X}} \left\langle \sum_{t=T-w+1}^{T} \sigma_t X_t, c \right\rangle \le 2 \mathop{\mathbb{E}}_{\sigma} \left\| \sum_{t=T-w+1}^{T} \sigma_t X_t \right\|$$
$$\le 2 \sqrt{\mathop{\mathbb{E}}_{\sigma}} \left\| \sum_{t=T-w+1}^{T} \sigma_t X_t \right\|^2 = 2 \sqrt{\sum_{t,t'} \langle X_t, X_{t'} \rangle \mathop{\mathbb{E}}_{\sigma} \sigma_t \sigma_{t'}}$$
$$\le 2 \sqrt{w} , \qquad (25)$$

where the first inequality is due to Jensen's inequality. By plugging (24) and (25) into the right-hand side of (23), we finally obtain that:

$$\hat{\mathcal{R}}_w(\overline{\mathcal{F}}) \le \frac{3z}{4}\sqrt{w} \quad . \tag{26}$$

The statement follows since inequality (26) holds for any choice of $\{X_{T-w+1}, \ldots, X_T\} \in \mathcal{X}^w$.

Proposition 17 (Lower bound to worst-case empirical Rademacher complexity) *The following inequality holds:*

$$ilde{\mathcal{R}}_w(\overline{\mathcal{F}}) \geq 2^{-3/2}\sqrt{w}$$
 .

Proof Without loss of generality, assume that w is even. Let e_1 be a canonical standard vector. We set half of the w points to be equal to e_1 , and the other half to be equal to $-e_1$, specifically $X_{T-w+1} = \ldots = X_{T-w/2} = e_1/2$ and $X_{T-w/2+1} = \ldots = X_T = -e_1/2$. We have that:

$$\begin{split} \tilde{\mathcal{R}}_{w}(\overline{\mathcal{F}}) &\geq \mathbb{E} \sup_{\sigma} \sum_{c \in \mathcal{C}} \left(\sum_{t=T-w+1}^{T-w/2} \sigma_{t} \| c - e_{1} \|^{z} + \sum_{t=T-w/2+1}^{T} \sigma_{t} \| c + e_{2} \|^{z} \right) \\ &\geq \mathbb{E} \max_{\sigma} \max_{c \in \{e_{1}, -e_{1}\}} \left(\sum_{t=T-w+1}^{T-w/2} \sigma_{t} \| c - e_{1} \|^{z} + \sum_{t=T-w/2+1}^{T} \sigma_{t} \| c + e_{2} \|^{z} \right) \\ &= \mathbb{E} \max \left(\sum_{t=T-w+1}^{T-w/2} \sigma_{t}, \sum_{t=T-w/2+1}^{T} \sigma_{t} \right) \; . \end{split}$$

Let $A = \sum_{t=T-w+1}^{T-w/2} \sigma_t$ and $B = \sum_{t=T-w/2+1}^{T} \sigma_t$. Note that $2 \max(a, b) = a + b + |a - b|$ for any $a, b \in \mathbb{R}$. Thus, we have that:

$$\tilde{\mathcal{R}}_w(\overline{\mathcal{F}}) \geq \mathop{\mathbb{E}}_{\sigma} \max(A, B) = \frac{1}{2} \mathop{\mathbb{E}}_{\sigma} \left| \sum_{t=T-w+1}^T \sigma_t \right| \geq 2^{-3/2} \sqrt{w}$$

where the last step is obtained by an application of Khintchine inequality (Garling, 2007).

Proof [Lemma 14] The proof follows as an immediate corollary of Lemma 15, Proposition 16 and Proposition 17. By Lemma 15, there exists a constant c > 0 such that for any set of samples $\{X_{T-w+1}, \ldots, X_T\}$, it holds that:

$$\hat{\mathcal{R}}_w(\mathcal{F}) \le c\sqrt{k}\tilde{\mathcal{R}}_w(\overline{\mathcal{F}})\log^2\left(rac{w}{\tilde{\mathcal{R}}_w(\overline{\mathcal{F}})}
ight)$$

By using Proposition 16, we upper bound $\tilde{\mathcal{R}}_w(\overline{\mathcal{F}}) \leq \frac{3z}{4}$. Additionally, the statement of Proposition 17 implies that $\sqrt{w}/\tilde{\mathcal{R}}_w(\overline{\mathcal{F}}) \leq \sqrt{w}2^{3/2}$. Thus, we can conclude that there exists a constant c' > 0 such that:

$$\hat{\mathcal{R}}_w(\mathcal{F}) \le c' z \sqrt{kw \log^4(w)}$$

Appendix C. Lower Bound (Theorem 4)

This section is devoted to the proof of the lower bound of Theorem 4. For ease of notation, we denote with $\mathcal{B}(0, r) = \{x \in \mathbb{R}^d : ||x|| \le r\}$ the ball centered at the origin with radius $r \ge 0$. To simplify the computations, without loss of generality, we assume that the supports of the distributions P_1, \ldots, P_T lie in a ball centered at the origin of radius 1 rather than 1/2.

Let $k \ge 3$. Without loss of generality, we assume that k is divisible by 3. Let m = k/3 and let z_1, \ldots, z_m be a $12\overline{\Delta}$ -net of $\mathcal{B}(0, 1-4\overline{\Delta})$ for some later defined $\overline{\Delta} > 0$. Let $w = (\overline{\Delta}, 0, \ldots, 0) \in \mathbb{R}^d$. We define the sets $U_{i,-1} = \{z_i - 3w, z_i - 2w\}$ and $U_{i,1} = \{z_i + 3w, z_i + 4w\}$ for $i = 1, \ldots, m$, and we let $U = \bigcup_{i=1}^m (U_{i,1} \cup U_{i,-1})$. By construction, the points U are contained in the unit ball $\mathcal{B}(0, 1)$. Following the volume argument of Levrard (2015), a sufficient condition on $\overline{\Delta}$ to guarantee that the points z_1, \ldots, z_m exist is

$$m \le \left(\frac{1-4\overline{\Delta}}{18\overline{\Delta}}\right)^d$$
,

and observe that $\overline{\Delta} = m^{-1/d}/18$ suffices.

Let $\delta = \Delta/(3\overline{\Delta})$ and let $r = (\frac{k}{48\delta^2})^{1/3}$. Let $\xi \in \{-1, 1\}^m$. We construct a sequence of discrete distributions $P_1^{\xi}, \ldots, P_T^{\xi}$ parameterized by ξ . Those distributions are defined as follows. If $t \leq T - r$, the distribution P_t^{ξ} is uniform over the points U:

$$P_t^{\xi}(x) = 1/(4m)$$
, $\forall x \in U_{i,\pm 1}, i = 1, \dots, m$

For t > T - r, the distribution P_t^{ξ} is drifting away from the uniform distribution with a drift structure described by ξ :

$$P_t^{\xi}(x) = \frac{1 \pm \xi_i (t - T + r)\delta}{4m} \qquad \forall x \in U_{i, \pm \xi_i}, \ i = 1, \dots, m$$

Proposition 18 For any ξ and $1 \le t < T$, it holds that

$$W(P_t^{\xi}, P_{t+1}^{\xi}) \leq \Delta$$
.

Proof If t < T - r, the Wasserstein distance is trivially equal to zero. If $T \ge T - r$, then we can transport the excess mass of P_{t+1}^{ξ} generated due to the drift in U_{i,ξ_i} to the set $U_{i,-\xi_i}$. With this transport, we obtain the following upper bound to the Wasserstein distance:

$$W(P_t^{\xi}, P_{t+1}^{\xi}) \le \sum_{i=1}^m \frac{2\delta}{4m} 6\overline{\Delta} = 3\overline{\Delta}\delta = \Delta$$
,

where the last equality is due to the definition of δ .

We denote with \tilde{C} the collection of all set of k centers $C \in C$ such that for each i = 1, ..., m, either:

- 1. C has centers $U_{i,1}$ and $z_i 5w/2$ (average of the points in $U_{i,-1}$)
- 2. C has centers $U_{i,-1}$ and $z_i + \frac{7w}{2}$ (average of the points in $U_{i,1}$).

For $\xi \in \{-1, 1\}^m$, we denote with $C^{\xi} \in \tilde{C}$ the solution such that if $\xi_i = 1$ (resp., $\xi_i = -1$), then the case (1) (resp., case (2)) above applies. Let $h(\xi, \xi')$ be the Hamming distance between two strings ξ and ξ' . The following proposition holds.

Proposition 19 *Let* $\xi, \xi' \in \{-1, 1\}$ *. Then:*

$$\mathcal{W}_{k,z}(C^{\xi}, P_T^{\xi'}) = \mathcal{W}_{k,z}^* \left(P_T^{\xi'} \right) + h(\xi, \xi') \frac{r \delta \overline{\Delta}^2}{4m}$$

Also, for any $C \in C$, there exists $\tilde{C} \in \tilde{C}$ such that for all $\xi \in \{-1, 1\}^m$, it holds that

$$\mathcal{W}_{k,z}(\tilde{C}, P_T^{\xi}) \leq \mathcal{W}_{k,z}(C, P_T^{\xi})$$
.

Proof The first statement can be immediately obtained through a direct computation. To prove the second statement, we can notice that the set $\mathcal{Z} = \{z_i \pm 3w : 1 \le i \le m\}$ contains points that are distance at least $6\overline{\Delta}$ from one another. As each point in U can be written as z' or z' + w for some $z' \in \mathcal{Z}$, Step 3 of Bartlett et al. (1998) shows that the statement is true if

$$6 \ge \sqrt{\frac{2}{1-r\delta}} + 1 \ .$$

It is possible to verify that this is indeed the case with our choice of r and δ .

Let $\mathcal{P}(\Delta)$ be the collection of all the sequence of joint distributions $P_1 \otimes \ldots \otimes P_T$ over $(X_1, \ldots, X_T) \subseteq \mathcal{B}(0, 1)^T$, such that $W(P_t, P_{t+1}) \leq \Delta$ for all $1 \leq t < T$ (Due to Lemma 7, this also implies an upper bound to the discrepancy τ .). For $\xi \in \{-1, 1\}^n$, consider the joint distribution $P^{\xi} = P_1^{\xi} \otimes \ldots \otimes P_T^{\xi}$ over (X_1, \ldots, X_T) , and observe that $P^{\xi} \in \mathcal{P}(\Delta)$ due to Proposition 18.

Let $\hat{C} \in \mathcal{C}$ be any solution that is decided based on the data (X_1, \ldots, X_T) . To prove the theorem, it is sufficient to show a lower bound to

$$\inf_{\hat{C}} \sup_{P_1 \otimes \ldots \otimes P_T \in \mathcal{P}(\Delta)} \mathbb{E} \left[\mathcal{W}_{k,z}(\hat{C}, P_T) - \mathcal{W}^*_{k,z}(P_T) \right] .$$

We have the following chain of inequalities.

$$\inf_{\hat{C}} \sup_{P_1 \otimes \ldots \otimes P_T \in \mathcal{P}(\Delta)} \mathbb{E} \left[\mathcal{W}_{k,z}(\hat{C}, P_T) - \mathcal{W}^*_{k,z}(P_T) \right] \\
\geq \inf_{\hat{C}} \sup_{\xi \in \{-1,1\}^m} \mathbb{E} \left[\mathcal{W}_{k,z}(\hat{C}, P_T^{\xi}) - \mathcal{W}^*_{k,z}(P_T^{\xi}) \right] \\
\geq \inf_{\hat{\xi}} \sup_{\xi \in \{-1,1\}^m} \mathbb{E} \left[\mathcal{W}_{k,z}(C^{\hat{\xi}}, P_T^{\xi}) - \mathcal{W}^*_{k,z}(P_T^{\xi}) \right] \\
= \frac{r\delta\overline{\Delta}^2}{4m} \inf_{\hat{\xi}} \sup_{\xi \in \{-1,1\}^m} \mathbb{E} \left[h(\hat{\xi}, \xi) \right] ,$$
(27)

where the last two steps are due to Proposition 19.

For two distributions P and Q, the square of their Hellinger distance is defined as

$$H^2(P,Q) = \int (\sqrt{P(x)} - \sqrt{Q(x)})^2 dx \quad .$$

When $P = P_1 \otimes \ldots \otimes P_T$ and $Q = Q_1 \otimes \ldots \otimes Q_T$, we have the following factorization property (Tsybakov, 2008):

$$H^{2}(\boldsymbol{P}, \boldsymbol{Q}) = 2 \left[1 - \prod_{i=1}^{T} \left(1 - \frac{H^{2}(P_{i}, Q_{i})}{2} \right) \right]$$
(28)

Proposition 20 Let $\xi, \xi' \in \{-1, 1\}^m$ be two strings that differ only in a single position, i.e. $h(\xi, \xi') = 1$. Then:

$$H^2(\boldsymbol{P}^{\xi}, \boldsymbol{P}^{\xi'}) \leq rac{1}{2}$$
.

Proof If $t \leq T - r$, then

$$H^2(P_t^{\xi},P_t^{\xi'}) = 0 \ ,$$

else if t > T - r, then

$$\begin{split} H^2(P_t^{\xi},P_t^{\xi'}) &= 4 \left[\sqrt{\frac{1+(t-T+r)\delta}{4m}} - \sqrt{\frac{1-(t-T+r)\delta}{4m}} \right]^2 \\ &= \frac{1}{m} \left[\sqrt{1+(t-T+r)\delta} - \sqrt{1-(t-T+r)\delta} \right]^2 \\ &\leq \frac{1}{m} [2(t-T+r)\delta]^2 \\ &= \frac{4r^2\delta^2}{m} \ , \end{split}$$

where the inequality is due to the fact that $\sqrt{x} - \sqrt{y} \le x - y$ if x > 1 and $0 \le y \le x$. By using the factorization property (28), we obtain

$$H^2(\boldsymbol{P}^{\boldsymbol{\xi}}, \boldsymbol{P}^{\boldsymbol{\xi}'}) \le 2\left[1 - \left(1 - \frac{4r^2\delta^2}{m}\right)^r\right] \le \frac{8r^3\delta^2}{m} \ .$$

The statement follows by substituting the values of r, δ and m.

We use Assouad's lemma as in Theorem 2.12 of Tsybakov (2008) to lower bound (27). Let $\alpha \doteq \sup_{\xi,\xi':h(\xi,\xi')=1} H^2(\mathbf{P}^{\xi}, \mathbf{P}^{\xi'})$. We have that if $\alpha \leq 2$, then

$$\inf_{\hat{\xi}} \sup_{\xi \in \{-1,1\}^m} \mathbb{E}\left[h(\hat{\xi},\xi)\right] \ge \frac{m}{4} (1 - \sqrt{\alpha(1 - \alpha/4)})$$

By using Proposition 20 in the above inequality and combining it with (27), we finally have that:

$$\inf_{\hat{C}} \sup_{P_1 \otimes \ldots \otimes P_T \in \mathcal{P}(\Delta)} \mathbb{E} \left[\mathcal{W}_{k,z}(\hat{C}, P_T) - \mathcal{W}^*_{k,z}(P_T) \right] \geq \frac{r \delta \overline{\Delta}^2}{48} .$$

We conclude the proof of the Theorem by substituting the values of r, δ and $\overline{\Delta}$.